

# Temi di discussione

(Working Papers)

Competition and the pass-through of unconventional monetary policy: evidence from TLTROs

by Matteo Benetton and Davide Fantino







# Temi di discussione

(Working Papers)

Competition and the pass-through of unconventional monetary policy: evidence from TLTROs

by Matteo Benetton and Davide Fantino

Number 1187 - September 2018

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Bassanetti, Marianna Riggi, Emanuele Ciani, Nicola Curci, Davide Delle Monache, Francesco Franceschi, Andrea Linarello, Juho Taneli Makinen, Luca Metelli, Valentina Michelangeli, Mario Pietrunti, Lucia Paola Maria Rizzica, Massimiliano Stacchini. Editorial Assistants: Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

# COMPETITION AND THE PASS-THROUGH OF UNCONVENTIONAL MONETARY POLICY: EVIDENCE FROM TLTROS

by Matteo Benetton\* and Davide Fantino\*\*

## Abstract

We make use of an allocation rule by the ECB for Targeted Longer-Term Refinancing Operations (TLTROs) to provide causal evidence on the effect of unconventional monetary policy on the cost of loans to firms. Using transaction-level data from Italy's Central Credit Register and a difference-in-difference identification strategy, we show that treated banks decrease loan rates to the same firm by approximately 20 basis points compared with control banks. We then study how the effects of the liquidity injection vary according to the competition in the banking sector, exploiting the local nature of bank-firm lending relationships and exogenous variations in the number of pawnshops across Italian cities during the Renaissance. Our results suggest that banks' market power can significantly impair the effectiveness of unconventional monetary policy, especially for safer and smaller firms.

JEL Classification: E51, E52, L11.

Keywords: unconventional monetary policy, bank competition, pass-through.

## Contents

1.	Introduction	5
2.	Institutional setting	10
	2.1 Targeted Longer-Term Refinancing Operations	10
	2.2 The Italian banking system	12
3.	Data	14
4.	Identification strategy	17
5.	Results	21
	5.1 The effect of targeted monetary policy	22
	5.2 The effect of competition on the transmission mechanism	23
	5.3 Heterogeneous effects across firms and banks	25
6.	Conclusions	27
Re	ferences	29
A.	Main figures and tables	34
B.	Additional figures and tables	43
C.	Additional features of TLTROs	48

\*Haas School of Business, University of California, Berkeley.

\*\*Bank of Italy, Directorate General for Economics, Statistics and Research.

# 1 Introduction<sup>1</sup>

Since the global financial crisis central banks around the world have implemented unprecedented measures to counteract the credit crunch and sustain economic activity, such as quantitative easing, liquidity injections and policy announcements. These new tools have spurred the academic and policy debate about the role of the banking sector for their transmission to the real economy (Gertler and Karadi, 2011; Chodorow-Reich, 2014; Acharya et al., 2015; Di Maggio et al., 2016; Rodnyansky and Darmouni, 2017; Agarwal et al., 2017; Krishnamurthy et al., 2017).

Empirical studies of the effects of both conventional and unconventional monetary policy on credit supply faces two well-known identification issues: simultaneous causality between credit demand and supply, and selection into treatment as banks choose to borrow from the central bank (Kashyap and Stein, 2000; Jiménez et al., 2012, 2014). Furthermore, disentangling the role of the banking sector in the transmission mechanism poses an additional identification challenge due to non-random assignment of banks' market power, which can be correlated in the cross-section with other confounding factors (Scharfstein and Sunderam, 2014; Drechsler et al., 2017; Agarwal et al., 2018).

In this paper we provide causal evidence on the effect of targeted unconventional monetary policy on banks' credit supply to firms and on the role of competition in the banking sector for the transmission mechanism. We study a series of Targeted Longer-Term Refinancing Operations (TLTROs) by the ECB announced on the 5th of June 2014 with the goal to enhance the functioning of the monetary policy transmission mechanism by supporting lending to the real economy. We exploit an allocation rule by the policy together with a rich dataset on transaction-level bank-firm lending relationships and with exogenous variation

<sup>&</sup>lt;sup>1</sup>We are grateful to Lorenzo Burlon, Luisa Carpinelli, Giovanni Favara, Alessandro Gavazza, Luigi Guiso, Michael Greenstone, Amir Kermani, Daniel Paravisini, Enrico Sette, Johannes Stroebel and Paolo Surico for helpful comments as well as seminar participants at LSE, Bank of Italy, ECB and Boston FED. The opinions expressed are those of the authors and do not necessarily reflect the views of the Bank of Italy or of the Eurosystem. All errors are our own.

in banks' local market power to address three identification challenges and shed light on the functioning of the transmission of unconventional monetary policy through the banking sector.

First, the dynamics of credit in the lending market are driven by both demand and supply. In equilibrium, borrowers' willingness to take new loans and accept different conditions as well as lenders' incentives to supply and reprice loans jointly determine the amount of credit and its price. Less risky borrowers may have a higher demand for the liquidity coming through TLTROs so that a decrease in lending rates comes from selection on the demand side, rather than from treatment on the supply side. To control for demand factors, we leverage on a panel of firms borrowing from multiple banks and estimate an empirical model with a full set of firm-time interacted fixed effects, thus only exploiting the variation within a firm across banks, as pioneered in Khwaja and Mian (2008).

A second identification challenge arises from selection into treatment. Even after controlling for demand factors, the supply side variation across banks that we use for identification may be endogenous because banks' use of TLTROs is a choice. To control for time-invariant bank level unobservables, we include in our empirical model bank fixed effect. However, timevarying differences across banks that affect both TLTROs borrowing and lending strategies, can still bias our results. We construct an instrumental variable (IV) for banks' treatment using a rule in TLTROs guidelines, that set the maximum amount that banks can borrow in the first two operations to 7% of their outstanding amount of eligible loans on April 2014. The threshold is set by the ECB for the whole euro area and is based on a variable that is fixed before the announcement of the policy. The differences in potential treatment across banks are therefore predetermined and orthogonal to unobservables than may affect loan supply in the period after TLTROs. The relevance of our instrument is ensured by the fact that in the first two TLTROs more than 90% of the banks actively participating to the operations borrowed at least 95% of their borrowing limit.

Third, to study how the transmission mechanism is affected by the structure of the

banking sector we need exogenous variation in competition. We define local banking markets the Italian provinces, the equivalent of US counties, and we assume that each firm borrows in the same province where it has its headquarter.<sup>2</sup> We isolate the effect of competition on the pass-through of unconventional monetary policy by exploiting geographical variation in banks' market shares across provinces. As the latter may be correlated with other factors affecting the equilibrium in the local credit market, we design an IV strategy using variation in the presence of pawnshops across Italian cities during the Renaissance as an instrument for the level of local competition in the banking sector today.<sup>3</sup>

Our first set of results looks at the effect of unconventional monetary policy on credit supply for firms. We find that banks participating to TLTROs decrease their rates to the same firm by 20 basis points relative to banks that do not participate, when we instrument banks' borrowing choice using the exogenous allocation rule. This effect is significant and represents approximately 5 percent of the baseline cost of credit. We allow the pass-through to vary over time and find that treated banks start decreasing rates about two quarters after the first liquidity injection. Our IV estimates are significantly larger than the OLS estimates, where banks choosing to borrow from TLTROs decrease rate by about 5 basis points relative to banks choosing not to borrow.

Our second set of results examines the role of banks' market power for the transmission mechanism of unconventional monetary policy. We find that competition plays a significant role for the pass-through of unconventional monetary policy, limiting the sensitivity of the cost of corporate loans to the cost of bank funding. The magnitude of the result is significant: a one-standard-deviation increase in concentration reduces the impact of TLTROS on

<sup>&</sup>lt;sup>2</sup>In our data we don't observe from which branch of the bank the firm borrows, but previous evidence for Italy (Bofondi and Gobbi, 2006; Felici and Pagnini, 2008; Crawford et al., 2018) and other countries (Petersen and Rajan, 1994, 1995; Mian, 2006) suggest that lending to firms has a local dimension. This choice is also motivated by the fact that provinces are the geographical units used by the regulator to approve branch openings. We use the structure of Italian provinces existing in 2005.

<sup>&</sup>lt;sup>3</sup>Monte dei Paschi di Siena, the world's oldest surviving bank, was founded in Siena by the city magistrates as a pawnshop in 1472. Guiso et al. (2004) and Pascali (2016) show the importance of historical difference in access to credit for long-term financial development across region and provinces in Italy.

lending rates by approximately 14 basis points. This corresponds to a 32% decline in the transmission of unconventional monetary policy relative to the benchmark of perfect competition. Furthermore, we find that in provinces with low concentration lenders pass-on the lower rates to borrowers immediately, while in provinces with high concentration banks do not lower rates immediately after the policy change, but start after two quarters.

Finally, we explore heterogeneous effects in the transmission mechanism of TLTROs due to differences in firms' and banks' characteristics. Small firms and those with better credit rating borrowing from a bank using TLTROs experience a decrease in the cost of credit, while the reduction is not significant for the other firms of the same bank. Banks' local market power affects the pass-through for smaller and safer firms, but plays no role for larger and riskier firms. The differential effect on small firms is consistent with previous studies showing that small firms have less alternatives than large firms in raising funding and may be more affected by bank' competition (Berger and Udell, 1995; Beck et al., 2004). The differential effect on ex-ante safer firms is consistent with theories based on information asymmetries and hold-up problems (Sharpe, 1990; Rajan, 1992). Our heterogeneity analysis suggests a flight-to-quality within the corporate sector, with large banks competing to allocate the ECB liquidity toward smaller and ex-ante safer firms, especially in more competitive provinces.

**Related literature** Our paper is related to two main strands of literature. First, we contribute to the empirical macroeconomic literature about the transmission mechanism of monetary policy and how this is affected by financial imperfections, by studying how unconventional monetary policy affect the cost of credit for firms with an innovative research design.<sup>4</sup> The key empirical challenge is to identify how monetary policy affects supply side-factors, when there are confounding demand side effects (Kashyap and Stein, 2000). A stream of literature has used firm-time fixed effects to controls for unobservable demand factors, together with exogenous measures of exposure to a shock. This approach has been

<sup>&</sup>lt;sup>4</sup>Theoretical works on the topic go back to the seminal contributions by Bernanke and Blinder (1992) and Bernanke et al. (1999). After the global financial crisis new models have included an active financial sector (Gertler and Kiyotaki, 2010; Brunnermeier and Sannikov, 2014) and studied its implications in a general equilibrium setting (Gerali et al., 2010; Gertler and Karadi, 2011).

adopted to study the effect of supply-side liquidity shocks (Khwaja and Mian, 2008; Schnabl, 2012; Gambacorta and Mistrulli, 2014), sovereign shocks (Bofondi et al., 2013; Albertazzi et al., 2014; De Marco, 2015), and the transmission mechanism of both conventional and unconventional monetary policy (Jiménez et al., 2012; Drechsler et al., 2016; Jiménez et al., 2014; Acharya et al., 2015; Carpinelli and Crosignani, 2017; Di Maggio et al., 2016). We are the first to study a new unconventional monetary policy which has been implemented with the explicit goal of increasing lending to the real economy, thus providing evidence on the value of setting a lending target for monetary policy effectiveness. Differently from most previous empirical studies that look at quantities our work focuses on the pass-through to interest rates, which have been less studied by the previous literature because of limited data availability on prices at the loan-level (Jiménez et al., 2014; Krishnamurthy et al., 2017). Most notably, in our setting we observe both actual and potential treatment based on an exogenous allocation rule, while previous studies generate cross-sectional variation in banks' exposure to a shock or a policy change using predetermined banks characteristics.

Second, our work contributes to the literature on the relation between competition and monetary policy. The industrial organization approach to banking literature has studied theoretically the link between competition and monetary policy (Freixas and Rochet, 2008; Rochet, 2009), but the empirical evidence about the relationship between market power and pass-through is ambiguous (Berger and Hannan, 1989; Neumark and Sharpe, 1992; De Graeve et al., 2007). On the one hand, in more competitive market the pass-through of borrowing rate to lending rates can be larger, as a results of higher elasticities of firms' loan demand and absence of smoothing coming from relationship lending (Cottarelli et al., 1995; Van Leuvensteijn et al., 2013). On the other hand, the response of lending rate can be higher in more concentrated market, if banks pass-through cost efficiency or exploit market power from holdup situations to adjust their markups (Petersen and Rajan, 1995). We develop a new identification strategy to study empirically the effect of competition on the transmission mechanism of unconventional monetary policy to corporate lending, complementing recent studies that look at the effect of competition for the transmission of monetary policy in the US mortgage and deposit markets (Scharfstein and Sunderam, 2014; Drechsler et al., 2017). Our focus on targeted unconventional monetary policy has the unique advantage that the treatment is by design heterogeneous across lenders, which allows us to separate differences across banks from differences in market structure.

The rest of the paper is organized as follows. Section 2 describes the institutional background of TLTROs and the Italian banking system; Section 3 summarizes the data; Section 4 explains the identification strategy; Section 5 presents our results and Section 6 concludes.

# 2 Institutional Setting

## 2.1 Targeted Longer-Term Refinancing Operations

On the 5th of June 2014, the ECB decided to support bank lending to the euro area non-financial sector through a first series of Targeted Longer-Term Refinancing Operations (TLTROs). This policy measure is implemented through eight auctions, one each quarter from September 2014 to June 2016, and participation is open to institutions that are eligible for the Eurosystem open market operations. In July 2014 and February 2015 the ECB updated the rules on borrowing limits, maturities and early repayment options. A second series of four operations starting in June 2016 has been announced on the 10th of March 2016.

The ECB has been actively involved in supporting the financial system since the onset of the global financial crisis in September 2008. In October 2008, the ECB switched to a fixedrate full-allotment mode for its refinancing operations, where the central bank sets an interest rate and banks can borrow an unlimited amount at that given rate. In this way the ECB provided a certain source of funding to banks, especially valuable in crisis time when other funding sources are impaired. The ECB also increased its support to the banking sector with Longer-Term Refinancing Operations (LTROs), complementing the weekly liquidityproviding transactions, that have usually a maturity of one to three months, with a one-year operation in July 2009 and two three-years operations in December 2011 and February 2012. This longer-term liquidity allows banks to relax the roll-over risk coming from the mismatch between assets and liabilities, thus favoring longer-term investment. The popularity of the two three-years LTROs is evident from banks' participation and take-up: these operations provided more than 1 trillion euros liquidity to euro area banks, with Spanish and Italian institutions among the main beneficiaries (Carpinelli and Crosignani, 2017). Banks used the provided liquidity for rolling over previous debt, issuing new loans to firms and household and buying sovereign bonds.

The TLTROs come within the framework of increasing support by the ECB, but with some novelties about both goals and rules. While previous operations were designed to support the banking sector, TLTROs explicitly target lending to the real economy. For this reason, this policy represents an ideal experiment to understand the full transmission mechanism from the central bank to firms and households, via the financial sector. Both the goals and the rules are implicitly designed to reduce the incentives to banks to use the liquidity for buying sovereign debt, as happened in previous operations (e.g. LTROs), and to roll over existing debt.<sup>5</sup>

Figure 1 shows the time-line of the first series of TLTROS. Participation to the operations was possible both on an individual basis and as a "TLTRO group" of banks, not necessarily all part of the same banking group. The individual institution and the "lead institution" in the group should be an eligible Eurosystem counterpart. The eligibility criteria, valuation, haircuts and rules on the use of assets for collateral are the same of the other standard refinancing operations (ECB, 2014). The interest rate on the TLTROS will be fixed over the life of each operation at the rate on the Eurosystem Main Refinancing Operations prevailing at the time of take-up; an additional fixed spread of 10 basis points has been added for the first two TLTROS.

The main differences of TLTROs' rules relative to previous operations are on borrowing

<sup>&</sup>lt;sup>5</sup>It is worth noting that the TLTROs overlap with the end dates of the previous LTROs, maturing on January 29, 2015 and February 26, 2015, and therefore part of the funds were anyway used to roll over the expiring debts of LTROs. For this reason in our estimation strategy we account for expiring debt.

limits. The borrowing limits rules are different for the first two operations at the end of September and December 2014 and the last six, from March 2015 to June 2016. Define  $q_k^b$ the quantity borrowed by bank *b* (single or "TLTRO group") in operation *k*. The initial borrowing limit for the first two operations is computed using the following formula:

$$q_b^1 + q_b^2 \le 0.07 \times EL_b^{April2014} \equiv Rule_b. \tag{1}$$

Bank *b* borrowing in the first two TLTROs cannot exceeds 7% of its outstanding amount of eligible loans on 30 April 2014  $(EL_b^{April2014})$ . The eligible loans include lending to domestic non-financial corporations and households in the euro area, and exclude loans securitised or otherwise transferred without derecognition from the balance sheet.<sup>6</sup> Moreover, they exclude loans to household for house purchases to emphasize even more the willingness of the ECB to channel new liquidity into productive investment. In Section 4 we describe how we use the rules regarding the borrowing limit for the first two TLTROs in our identification strategy, while in Appendix C we describe additional rules of the scheme for the last six operations and repayments.

## 2.2 The Italian Banking System

The supply of bank credit is particularly important in Italy as firms are heavily dependent on intermediated credit, relative for example to U.S. firms (Langfield and Pagano, 2016). Italian banks have traditional business models, based on loans to the real economy and close relationship with their customers, through a developed network of branches. Guiso et al. (2004) report that "The president of the Italian Association of Bankers (ABI) declared in a conference that the banker's rule-of-thumb is to never lend to a client located more than three miles from his office" and they show how distance continue to segment local markets. Between 2008 and 2013 the number of branches decreased by 7% from 34, 100 to 31, 700, mostly as a results of large groups reorganizations. Despite this reduction in the network, the

 $<sup>^{6}</sup>$ The definitions are detailed in ECB (2014).

number of banks' employees working in local branches is stable at 65% and a survey of senior executives of the main Italian banks reveal that business originations through branches will continue to play a leading role together with online banking (PwC, 2010). In our analysis we consider a province as the relevant market for banks lending to firms. Provinces are geographical entities very similar to U.S. counties and they are used by the Italian antitrust authority as proxies for the local markets for deposits (Bofondi and Gobbi, 2006; Felici and Pagnini, 2008; Crawford et al., 2018). Figure 2 shows the geographical distribution of the quartiles of the Herfindahl index (HI) across provinces calculated on the outstanding amounts of the term loans in the first quarter of 2014. It shows that, even if competition is slightly stronger in the north-east, generally there is a lot of variability among geographically neighboring provinces.<sup>7</sup>

Italian banks' funding has experienced significant changes during the European sovereign crisis. With respect to short-term funding, retail deposits remained a stable source for Italian banks, while short-term wholesale funding was affected by a widespread flight-to-quality from peripheral to core countries. Long-term unsecured wholesale funding became increasingly harder to obtain for Italian banks, which restored to secured long-term funding via covered bonds. The rating of the debt issued have deteriorated, mostly as a results of the increase in non-performing loans, due to a fall by 9% of GDP and 25% of industrial production. These losses impacted negatively on Italian banks' capital and together with the deleveraging needed to improve capital ratio, severely reduce the capacity to provide loans to the real economy. In this context, central bank liquidity become increasingly more important as a source of funding for banks. The reliance of Italian banks on ECB funding, measures as a percentage of assets, grew from less than 1% at the end of 2010 to more than 6% at the end of 2012 (Van Rixtel and Gasperini, 2013). The new TLTROs by the ECB strengthen this trend, by providing additional long-term liquidity to banks in the euro area, with the explicit goal of promoting loan to firms. In the first two TLTROs, the banks of the euro area

<sup>&</sup>lt;sup>7</sup>The main exception here is Sardinia, that being a relatively distant island from the mainland suffers from isolation.

borrowed collectively 212 billion euros, with Italian institutions in the first place borrowing 57 billion euros. The transmission of TLTROs could have therefore important implications for lending to the Italian economy. The local and bank-centered Italian loan market and the importance of the ECB TLTROs for the liquidity of Italian banks' make our environment particularly suitable to investigate the transmission mechanism of monetary policy to the real economy.

# 3 Data

In this work we construct a unique dataset at the bank-firm-time level, combining four different sources of data.<sup>8</sup> The main one is the Italian Credit register, which collects individual data on borrowers with exposure above 30 thousand euros from all the intermediaries operating in Italy. From this source we extract information at a monthly frequency about the interest rates of term loans charged on bank debt for each borrower. Each observation is a bank-firm pair and we observe a unique identifier for both the lending and the borrowing institution. We collapse the data at the level of firm-banking group relationship using the mapping from the Supervisory register of the Bank of Italy, where the legal structure of all the Italian banking groups is publicly available.

We complement this data with additional information from both the bank and the borrower side. On the one hand, we collect quarterly data on the geographical distribution of branches and the structure of its balance sheet for each bank from the confidential Supervisory reports and the Supervisory register of the Bank of Italy. On the other hand, we exploit the borrower identifier to add information on the geographical location, the credit quality and the size of the firm, matching our dataset with the Company Accounts Data Service (CADS) managed by Cerved, one of the most comprehensive sources of information about balance sheets of Italian firms, also used by banks for credit decision. A last piece

<sup>&</sup>lt;sup>8</sup>We use the term bank to indicate both standalone banks and banking groups henceforth. For banks belonging to a banking group we aggregate the data at the banking group level, which is the relevant entity for borrowing from the ECB.

of information includes confidential data about participation and the amounts lent from the central bank to the Italian banks after each TLTRO bid.

The final dataset is a quarterly balanced panel, in a time span between the start of 2014 and the second quarter of 2015. Table 1 shows the summary statistics of the variables of the dataset. Panel A shows the main dependent variable of the analysis is the overall interest rate ( $r_{bft}$ ), including the accessory expenses, on the stock of term loans, charged by bank b to firm f at time t, shown in panel A of the table. The first and the last percentile of the distribution of the interest rates have been winsorized, to minimize the impact of outliers in the sample. The charged interest rate has been equal on average to about 4% considering the whole time series and the whole distribution is included between about 0.5% and 12%. For this kind of loans the impact of the expenditures on the overall rate is not particularly strong. We also show the statistics on the interest rate on the flow of new term loans of each period, used in a robustness check, which are not substantially different from those on the stocks.

Panel B shows the Herfindahl Index of the local term loans markets. The first one is calculated using quantities of credit for each province in the first quarter of 2014.<sup>9</sup> The credit market in Italy is relatively concentrated, with an average value of the index of 0.17 and a range of values included between 0.09 and 0.36. In section 5.3 we assume in an extension of our empirical model that markets are segmented according to the credit quality of the borrower, summarized in nine ordinal categories by an index of credit riskiness taken from CADS and calculated from the available balance sheet data. We construct for this exercise separate HI assuming that the market of credit for firms of average and high credit quality (classes 1-6 of the rating index) is different from the one for firms of low quality (classes 7-9).<sup>10</sup> The statistics for the HI of the two segmented markets are very similar both among them and to those of the HI by province. We also show summary statistics for the total

 $<sup>^9 \</sup>rm We$  used the structure of 103 Italian provinces existing until 2005 to get a homogeneous classification of the provinces from the different datasets.

<sup>&</sup>lt;sup>10</sup>We also segmented the credit market in three categories instead of two, but the final results were the same as those presented here.

number of pawnshops that opened during the Italian Renaissance across Italian province that we compute aggregating the city level data from Pascali (2016). The average number of pawnshop by province is about one and it ranges from zero to eight.

Panel C shows the variables regarding the first two TLTROs; 78 banks in our sample participated at either the first or the second TLTRO and the average borrowed amount was about 670 million euros. Several of those banks used anyway either part or all the borrowed liquidity to rollover already existing debts with the Eurosystem; for this reason we calculated a corrected amount of the exposition to central bank coming from the first two TLTROs, netting out the debts towards the Eurosystem expiring in the same quarter. 43 banks had a positive net amount after this correction and the average net borrowed amount was of about 550 million euros; the distribution is skewed to the left and the range of values is included between 5 and about 5500 million euros. The borrowing limit for the first two TLTROs, calculated for the whole sample of 104 banks was on average of about 550 million euros too, but is more skewed to the left. From the comparison of the raw amount borrowed from the 78 banks participating to the TLTROs with their borrowing limit we find that more than 90% of those banks borrowed more than 95% of their limit.

In panel D we report the main structural characteristics of the banks in the first quarter of 2014: they had on average 30 billion euros of assets, almost half of which are loans and about 20% are government bonds. The riskiness of the credit portfolio of the banks and capital adequacy are respectively measured by the ratio between bad loans and overall loans, equal on average to about 9%, and by the capital ratio, based on the Basel rules, equal on average to about 15%.

Last, in panel E we report some statistics regarding firm characteristics, taken from the balance sheets in CADS for the year preceding the policy (2013) and used in the analysis of heterogeneous effects in Section 5.3. We show the distribution of firm assets, equal on average to 4 million euros, and the percentages of firms whose credit quality is either high/average (about three quarters of the sample) or low. We compare the statistics where the statistical unit is the firm with those where the statistical unit is the firm-bank relationship (which is

the relevant statistical unit in our final dataset); the statistics are substantially similar in both cases, taking into account that on average bigger firms have more credit relationships and therefore their weight is bigger when the statistical unit is the relationship.

In Table 2 we compare the characteristics of the banks borrowing a positive additional amount of resources from the TLTROs and of their customer firms (treated group) with the other banks and firms (control group). Panel A shows the existing differences in the endogenous variables in the first three quarters of 2014, between relationships of firms with a treated bank and the other relationships. We do not find appreciable differences in the statistics. In panels B and C we check the borrowing limit and the structural characteristics for treated and control banks. We find that on average the former are bigger than the latter and have therefore a bigger borrowing limit, but they are substantially similar when checking for the other characteristics. In panel D we contrast the statistics weighted by the number of firm-bank relationships of the firms borrowing respectively from a treated or a control bank; also in this case we do not find evidence of relevant differences in the two samples.

# 4 Identification Strategy

First, to study the effect of unconventional monetary policy on the cost of credit, ideally one would randomly assign liquidity to identical banks lending to the same firm. Any decrease in the lending rate from the bank receiving the liquidity will come from the treatment and not from other banks characteristics (the two banks are identical) or firms characteristics (they are lending to the same firm). Second, to study the effect of bank competition on the pass-through of unconventional monetary policy, ideally one would like that the bank receiving the random liquidity injection operates in two identical markets, which differ only on the level of bank competition. Any differential decrease in the lending rate from the bank receiving the liquidity in the market with high bank competition will come from the differences in bank competition and not from other banks characteristics (it is the same bank) or other markets characteristics. Our empirical strategy proceeds in three steps to address the key identification challenges: simultaneous causality, selection into treatment and omitted variables.

Simultaneous causality. TLTROs were designed and implemented by the policymaker as a reaction to macroeconomic conditions to explicitly promote lending to the real economy. Therefore, macroeconomic shocks correlated to the policy may induce unobservable loan demand shifts that are contemporaneous to the ECB interventions, leading to simultaneity and omitted variables bias. An upward bias in the evaluation of the effects of the policy would result if safer firms demand from banks borrowing from TLTROs; while a downward bias would emerge if riskier firms increase their loan demand by more. To control for changes in lending opportunities we include in our specification interacted firm-time fixed effects. In this way, we capture firm-specific time-varying shocks to loan demand and we exploit only the variation within each firm-time pair across banks for identification.

We address possible concerns about differences at bank level controlling for time-invariant unobserved heterogeneity with bank fixed effects. In this way we capture, among other things, constant differences across banks in lending strategies and funding costs and we exploit only variation within bank over time. Moreover, we include time-varying bank controls (bank capital, non-performing loans, government bonds), that can have an effect on both banks' funding costs and borrowing decisions and are exogenous with respect to the rate decisions regarding a single transaction.

We estimate a difference-in-differences model on a balanced panel of firm-bank relationships.<sup>11</sup> We include in our equation time varying coefficients to capture the dynamics of the transmission mechanism and we cluster the standard errors both by firm and by bank-time. Hence, the resulting OLS empirical specification is:

$$Y_{bfmt} = \sum_{\tau} \alpha_{\tau} \mathbb{I}_{\tau=t} \times TLTRO_{b\tau} + \gamma_{ft} + \gamma_b + \theta X_{bt} + \varepsilon_{bfmt},$$
(2)

<sup>&</sup>lt;sup>11</sup>The use of the difference-in-differences methodology on a balanced panel implies that our conclusions only regards the credit relationships already existing before the start of the policy since the beginning of the pre-treatment period in the dataset (first quarter of 2014) and whose existence continued until the second quarter of 2015.

where  $Y_{bfmt}$  is the loan rate from bank b to firm f in market m and period t;  $TLTRO_{bt}$  is the treatment variable;  $\gamma_{ft}$  are firm-time fixed effects;  $\gamma_b$  are bank fixed effects and  $X_{bt}$  are time varying banks controls.

Selection into treatment. Even controlling for endogenous timing of TLTROs, participation is on a voluntary basis, within the rules set by the ECB and described in Section 2. This may add additional selection bias, due to non-random treatment assignment: the evaluation of the policy may be biased upward if banks with higher return to lending or lower funding costs *choose* to borrow more, or biased downward if banks with unobservable funding problems or lower marginal propensity to lend exploit more the ECB facilities. We explicitly address this self-selection problem, exploiting the institutional setting of the policy: we instrument *actual* borrowing for the first two TLTROs with the *maximum* borrowing limit rule described in equation (1) in Section 2. Our first stage regression of *actual* participation (*TLTRO*<sub>bt</sub>) on the exogenous regressors and the excluded instrument is:

$$TLTRO_{bt} = \phi Rule_b \times Post_t + \gamma_{ft} + \gamma_b + \theta X_{bt} + \epsilon_{bfmt}, \tag{3}$$

where  $TLTRO_{bt}$  is the actual treatment variable;  $Rule_b$  is the allocation rule for bank b from equation (1) and  $Post_t$  is a dummy equal to one after the implementation of the TLTROS. The borrowing limit has been set by the ECB in its announcement in June 2014 and it is based on an exogenous parameter, which is common across banks, and pre-determined banks' balance sheet characteristics.<sup>12</sup> The identifying assumption is that the borrowing limit established by the ECB for the first two TLTROS is a valid instrument for bank access to central bank liquidity controlling for unobservable time-varying demand heterogeneity (firm-time fixed effects), unobserved bank heterogeneity (bank fixed effects) and time-varying

 $<sup>^{12}</sup>$ We find a correlation of -0.007 between the loan-level interest rate in the pre-treatment period and the borrowing limit, suggesting that the borrowing limit is essentially uncorrelated with the dynamics of the cost of credit before the treatment.

bank characteristics. The resulting IV empirical specification is:

$$Y_{bfmt} = \sum_{\tau} \alpha_{\tau} \mathbb{I}_{\tau=t} \times T \widehat{LTRO}_{b\tau} + \gamma_{ft} + \gamma_b + \theta X_{bt} + \varepsilon_{bfmt}, \tag{4}$$

where  $T\widehat{LTRO}_{bt}$  is the predicted participation and all other variables are as in equation (2).

**Omitted variables.** In the second part of the paper we study the role of the local banking system for the pass-through of TLTROS. To identify the effect of competition among lenders on the pass-through of unconventional monetary policy, we exploit variation in the competitive structure at the local geographical level. We measure competition with the HI for corporate loans in the province where a firm headquarter is located  $(HI_m)$ , as Figure 2 shows. We augment equation (4) with time varying coefficients on the interaction between the treatment and the HI, to capture the dynamic effect of market power on the transmission mechanism:

$$Y_{bfmt} = \sum_{\tau} \alpha_{\tau} \mathbb{I}_{\tau=t} \times T \widehat{LTRO}_{b\tau} + \sum_{\tau} \beta_{\tau} \mathbb{I}_{\tau=t} \times T \widehat{LTRO}_{b\tau} \times HI_m + \gamma_{ft} + \gamma_b + \theta X_{bt} + \varepsilon_{bfmt}.$$
(5)

Variation in the HI can be correlated with other factors affecting the pass-through of unconventional monetary policy.<sup>13</sup> To account for endogeneity in market power we also instrument the HI using exogenous variation in the presence of pawnshops across Italian cities during the Renaissance. Pascali (2016) shows that variation in the presence of Jewish communities and pawnshops during the Italian Renaissance is correlated with the variation in financial development across Italian cities today. We exploit the same historical variation to instrument for the level of competition in the banking sector today. Figure 2 shows the distribution of HI today across Italian provinces, while Figure 3 shows the number of pawnshops during the Renaissance in the currently established provinces. From a graphical inspection of the two maps we see that provinces with a high number of pawnshops during

<sup>&</sup>lt;sup>13</sup>For example Beraja et al. (2017) show how the time-varying regional distribution of housing equity influences the aggregate consequences of monetary policy through its effects on mortgage refinancing.

the Renaissance tend to have a less concentrated banking sector today. The correlation coefficient is -0.27 and we formally test the relevance of our instrument with the first stage regression:

$$TLTRO_{bt} \times HI_m = \phi Rule_b \times Post_t \times Pawnshop_m + \gamma_{ft} + \gamma_b + \theta X_{bt} + \epsilon_{bfmt}, \qquad (6)$$

where  $Pawnshop_m$  is the number of pawnshops across Italian provinces during the Renaissance. To capture jointly the causal effects of unconventional monetary policy and of competition on the transmission mechanism, we estimate the following IV empirical specification:

$$Y_{bfmt} = \sum_{\tau} \alpha_{\tau} \mathbb{I}_{\tau=t} \times T \widehat{LTRO}_{b\tau} + \sum_{\tau} \beta_{\tau} \mathbb{I}_{\tau=t} \times T LT \widehat{RO}_{b\tau} \times H I_m + \gamma_{ft} + \gamma_b + \theta X_{bt} + \varepsilon_{bfmt},$$
(7)

where  $TLTRO_{b\tau} \times HI_m$  is the predicted interaction between the policy and the HI from the first stage regression of the actual interaction on the exogenous variables and the excluded instruments (the allocation rule and the presence of pawnshops during the Renaissance). The interaction term  $TLTRO_{bt} \times HI_m$  captures the causal effect of competition on the pass-through of unconventional monetary policy.

# 5 Results

In this section we describe our results. In Section 5.1 we show our first set of results on the effect of targeted monetary policy on the cost of credit. In Section 5.2 we discuss our second set of results on the role of bank competition for the transmission of targeted monetary policy. In Section 5.3 we study how the transmission mechanism varies with firms' and banks' characteristics, namely credit-risk and size.

## 5.1 The Effect of Targeted Monetary Policy

The first empirical result of interest is the identification of the causal effect of targeted monetary policy on the dynamics of the overall cost of credit. We estimate both one specification where the TLTRO variable is a dummy equal to one after the start of the policy if the institution participates in one of the first two operations and another specification where we use the log of the actual additional borrowed amount measuring the intensity of treatment.

Table 3 presents our results for the OLS model. Column (1) shows the results for the full sample in which we control for bank, firm and time fixed effects separately. Treated banks decrease interest rates relative to control banks, but the effects are not significant. In column (2) we estimate the OLS model on the sample of firms with multiple banking relationships and control for demand with interacted firm-time fixed effects. The effects are stronger and marginally significant. Banks borrowing from the ECB through TLTROs decrease lending rate relative to banks not borrowing by approximately 3 basis points on average. Finally in column (3) we add time-varying banks' control. The results are stronger and marginally more significant. In the last three columns of Table 3 we estimate the same model with the actual amount borrowers from the ECB, rather than the binary participation dummy. The results are similar to the ones with the binary treatment.

Table 4 presents our results for the IV model.<sup>14</sup> Column (1) shows the results for the full sample in which we control for bank, firm and time fixed effects separately, and we instrument the binary TLTROs participation with the ECB allocation rule. When we instrument the actual amount borrowed in the first two TLTROs in column (1) of Table 4, we find that treated banks decrease interest rates relative to control banks. Most notably, we find a statistically significant negative coefficients in the first and second quarter of 2015, therefore just after the implementation of the second round of the policy.

<sup>&</sup>lt;sup>14</sup>In Appendix B we show the first stage regression for the benchmark case with interacted firm-time fixed effects. *Predicted* TLTROs participation has a significant positive effect on *actual* TLTROs participation. The overall Kleibergen-Paap F-statistics for the first stages, approximately 42 and 56 for dummy and continuous treatments respectively, are well above the 10% Stock and Yogo (2002) weak identification test critical values of about 16.

In column (2) of Table 4 we estimate the benchmark case with interacted firm-time fixed effects, to capture differences in firms credit demand. Treated banks decrease rate to the same firm on average by about 23 basis point relative to control banks in the first and second quarter of 2015. Finally, in column (3) we control for time varying bank factors that can affect differentially the pricing within firm-time across banks. The results are still significant and the magnitude is reduced to about 20 basis points. A comparison of columns (3) from Tables 3 and 4 shows that our IV estimates are stronger than the OLS estimates, suggesting that unobservable heterogeneity is likely to bias our estimates downward. For example banks choosing to borrow from the ECB may have been the ones planning to lower corporate rates for other reasons (e.g. business strategy), that can be correlated with the choice to borrow from the ECB in the first place.

Overall, our results on price suggest an outward shift after the second TLTRO in the supply of loan by banks exploiting the liquidity injection by the ECB. In the next section we further corroborate this hypothesis and explore if the competitive environment has an effect on the pass-through of unconventional monetary policy.

## 5.2 The Effect of Competition on the Transmission Mechanism

Our second set of results shows how bank competition affects the transmission mechanism of TLTROs to the cost of credit. Table 5 presents the results. Also in this section, we focus on firms with multiple lending relationship, to isolate a credit supply shock, and control for differences across banks with both banks' fixed effects and time-varying controls. The effect of the instrumented TLTROs treatment broadly confirm the results from Table 4: banks exploiting the ECB liquidity injections decrease loan rates for firms more than banks not participating to TLTROs.

The coefficients of interest capture the interaction between TLTROs treatment and bank competition, measured by the local HI. Our estimates in column (1) of Table 5 imply that high concentration reduce the pass-through of unconventional monetary policy to firms through the cost of credit. In markets with an average level of concentration treated banks pass on the lower rates to borrowers immediately after the treatment. In markets with higher level of concentration treated banks do not lower rates immediately after the policy change, but start after two quarters. This effect may be due to second round effects following the reactions of other competitors in the market.

We find that competition plays a significant role for the pass-through of ECB liquidity on the cost of credit. The magnitude of the result is also significant: a firm in a province with a standard deviation higher level of concentration experiences a 14 basis points lower decline in the cost of credit. Higher concentration reduces the transmission mechanism of unconventional monetary policy to firms by approximately 32% relative to the theoretical case of perfect competition.<sup>15</sup> Our estimates of the effect of competition on lending rates are slightly larger than the ones from recent works on the pass-through of monetary policy to mortgage and deposits rates (Scharfstein and Sunderam, 2014; Drechsler et al., 2017). There are many possible reasons for this difference. First, relationship lending, information frictions and market power may play a more important role for corporate lending than for mortgage lending and bank deposits, in which products are more standardized. Second, the type of policy we are looking at. Both Scharfstein and Sunderam (2014) and Drechsler et al. (2017) focus on the transmission of conventional monetary policy, while we look at targeted monetary policy operations. Third, our identification strategy differs in how we control for banks' lending opportunities.

In column (2) of Table 5 we show the estimates of equation (7), thus instrumenting for both participation to TLTROs and local competition with the number of pawnshop in the same province during the Italian Renaissance.<sup>16</sup> Column (2) of Table 5 shows that our results on the competition channel are robust to confounding factors at the market level. Higher concentration in the local banking market, coming from exogenous historical

 $<sup>^{15}</sup>$ We compute the effect using the estimates from column (1) of Table 5 and taking the average of the effect of a standard deviation increase in concentration in each period.

<sup>&</sup>lt;sup>16</sup>In Appendix B we show the first stage estimate for our IV strategy, when we instrument both the TLTROs treatment and the interaction term TLTROs  $\times$  HI. Our instruments are significant and have the expected sign. The overall Kleibergen-Paap F-statistics for the first stages for both dummy and continuous treatments and both endogenous variables are well above the 10% Stock and Yogo (2002) weak identification test critical values of about 7.

variation, significantly reduces the pass-through of central bank liquidity to lending rates to firms, confirming our baseline result. Finally in columns (3) and (4) of Table 5 we report the estimates using a continuous treatment variable and the results are robust.

In Appendix B we show several robustness checks. First, we replicate the analysis using the raw amounts borrowed by banks in the TLTROs instead of the additional amount net of the rolled over already existing debts towards the Eurosystem. The results are qualitatively similar to the previous ones. Second, we reply the analysis for rates constructed with interest expenditures only, excluding accessory expenses from the calculation. As expected, the results are unaffected. Third, we show the results considering the interest rate on the flows of new loans of the period instead of the one on the overall stock. On one hand flows allow to better capture the dynamics of the new credit loans period by period, on the other hand they are less suitable than stocks to construct a representative balanced panel because the firm would need to borrow a new amount of credit in each period to be included in the sample. When considering the direct effect only, the results are stronger than for stocks and it is already statistically significant at the end of 2014, even if weaker than in the following quarters. The larger magnitude and significance can be explained by the fact that we are now only focusing on the new credit contracts agreed in each period and not on the overall stock of loans already agreed. When including the interaction with competition, the results are qualitatively similar to those for stocks, even if not strongly significant as the results of Table 5 because of the loss of precision in the estimates due to the smaller number of observations.

## 5.3 Heterogeneous Effects Across Firms and Banks

In this section we study whether there are heterogeneous effects in the pass-through of unconventional monetary policy and the impact of the competitive environment due to differences in some relevant banks' and firms' characteristics. In particular, we focus on riskiness and size, which the previous literature identified as important determinants of access to credit (Jiménez et al., 2014; Agarwal et al., 2018). In both cases we take ex-ante measures of credit risk and size, to deal with possible endogeneity concerns.

Table 6 shows the estimates of model (4) in the different subgroups.<sup>17</sup> Columns (1) and (2) focus on firms' credit risk. We assume that credit markets are segmented by credit rating of the firm and calculated a different HI separately for each group of firms. We split the full sample into two subgroups: firms with good or average credit rating (classes 1-6) and those with a bad one (7-9).<sup>18</sup> We find that the reduction in the cost of credit is driven by loans to safer firms, while we do not find significant reduction in the cost of credit for riskier firms. Moreover, competition affects the pass-through of unconventional monetary policy to safer firms, but plays no role for riskier firms. This result corroborates the hypothesis that banks using the ECB facility to compete for the safest borrowers, as proxied by their ex-ante riskiness, while there is less space for competition in riskier lending.

Columns (3) and (4) of Table 6 look at differences in the pass-through between large and small firms. Here we split the sample taking firms above and below the median of the distribution of assets in the pre-treatment year (2013). We find that both groups benefit from the reduction in the cost of credit following the first two TLTROs, but the effect is stronger and only significant for smaller firms. Banks taking the ECB facility lowered the cost of credit to small firms by about 60 basis point, while the decrease is about 30 basis points smaller and not significant for large firms. Competition affects the pass-through of policy to the cost of credit for small firms, while it plays no significant role for large firms. This result is consistent with the idea that small firms benefit more from competition between lenders, because they have less alternatives than large firms in raising funding (Berger and Udell, 1995; Beck et al., 2004).

Finally, in columns (5) and (6) of Table 6 we study heterogeneity on the supply side and compare the largest five banks in Italy with other medium and co-operative banks. Treated large banks decrease their lending rate relative to control banks, while we do not

 $<sup>^{17}</sup>$ In Appendix B we report the estimates for the same specification using the continuous TLTROs measure as treatment variable. Results are confirmed.

 $<sup>^{18}</sup>$ We also considered a sample split in three categories (1-3) (4-6) (7-9) and the results were very similar to the split in two groups presented here.

find significant differences when the treated bank is of smaller size. Competition affects how large is the banks' pass-through to lending rates of the ECB liquidity. We find a positive significant interaction between the policy and the HI. Large banks decrease lending rate as a response to the lower funding cost in markets where they face competition from other banks, while they increase profit margins in markets where they have market power.

# 6 Conclusions

In this paper we empirically study the transmission mechanism of unconventional monetary policy to lending to firms and how it is affected by banks' market power. We exploit a rule set by the ECB on banks' borrowing limit as an instrument to identify an exogenous expansion in banks' funding availability, together with rich transaction-level dataset on term loans bank-firm lending relationship and exogenous historical variation in competitiveness of local lending market.

We show three main new findings. First, banks participating to the first two TLTROS decrease on average loan rates to the same firm by approximately 20 basis points relative to banks not participating to the ECB liquidity injection. Second, competition in the banking sector plays a significant role for the pass-through of TLTROS on the cost of credit: a one standard deviation increase in concentration reduces the decline in the cost by about 14 basis points, thus lowering the effect of unconventional monetary policy by approximately 32% relative to a perfect competition benchmark. Third, our effects are driven by large banks passing-through the ECB liquidity injection via lower loan rates to smaller and ex-ante safer firms, especially in more competitive markets.

Our results have important implications for both the implementation of monetary policy and the design of regulation to promote competitiveness in lending markets. Our analysis suggests that targeted monetary policy could be an effective tool for channeling banks funding into productive investment such as corporate lending, potentially avoiding unintended consequences (Acharya and Steffen, 2015; Crosignani et al., 2017). However, variation in banking competition changes the effects of monetary policy, potentially amplifying pre-existing differences in local credit access and economic conditions. We leave a more thorough analysis of the effects for the real economy to future work, but our results suggest that it is important for policy makers to consider the interactions between monetary and competition policies, especially following the recent changes in the competitive landscape due to consolidations, branch closures and the rise of shadow banks (Buchak et al., 2017; Stackhouse, 2018).

# References

- Acharya, Viral V., Tim Eisert, Christian Eufinger, and Christian Hirsch (2015) "Whatever It Takes: the Real Effects of Uncoventional Monetary Policy," Unpublished working paper.
- Acharya, Viral V. and Sascha Steffen (2015) "The 'Greatest' Carry Trade Ever? Understanding Eurozone Bank Risks," *Journal of Financial Economics*, Vol. 115, pp. 215–236.
- Agarwal, Sumit, Gene Amromin, Itzhak Ben-David, Souphala Chomsisengphet, Tomasz Piskorski, and Amit Seru (2017) "Policy Intervention in Debt Renegotiation: Evidence from the Home Affordable Modification Program," *Journal of Political Economy*, Vol. 125, pp. 654–712.
- Agarwal, Sumit, Souphala Chomsisengphet, Neale Mahoney, and Johannes Stroebel (2018)
  "Do Banks Pass Through Credit Expansions to Consumers Who Want to Borrow?," The Quarterly Journal of Economics (Forthcoming).
- Albertazzi, Ugo, Tiziano Ropele, Gabriele Sene, and Federico M. Signoretti (2014) "The Impact of the Sovereign Debt Crisis on the Activity of Italian Banks," *Journal of Banking & Finance*, Vol. 46, pp. 387–402.
- Beck, Thorsten, Asli Demirguc-Kunt, and Vojislav Maksimovic (2004) "Bank Competition and Access to Finance: International Evidence," *Journal of Money, Credit and Banking*, Vol. 36, pp. 627–648.
- Beraja, Martin, Andreas Fuster, Erik Hurst, and Joseph Vavra (2017) "Regional Heterogeneity and Monetary Policy," NBER Working Paper No. 23270.
- Berger, Allen N. and Timothy H. Hannan (1989) "The Price-Concentration Relationship in Banking," The Review of Economics and Statistics, Vol. 71, pp. 291–299.
- Berger, Allen N. and Gregory F. Udell (1995) "Relationship Lending and Lines of Credit in Small Firm Finance," *Journal of Business*, Vol. 68, pp. 351–381.
- Bernanke, Ben S. and Alan S. Blinder (1992) "The Federal Funds Rate and the Channels of Monetary Transmission," The American Economic Review, Vol. 82, pp. 901–921.

- Bernanke, Ben S., Mark Gertler, and Simon Gilchrist (1999) "The Financial Accelerator in a Quantitative Business Cycle Framework," in John B. Taylor and Michael Woodford eds. *Handbook of Macroeconomics*, Vol. 1C: Elsevier, pp. 1341–1393.
- Bofondi, Marcello, Luisa Carpinelli, and Enrico Sette (2013) "Credit Supply During a Sovereign Debt Crisis," *Journal of the European Economic Association*, Vol. 115, pp. 215–236.
- Bofondi, Marcello and Giorgio Gobbi (2006) "Informational Barriers to Entry into Credit Markets," *Review of Finance*, Vol. 10, pp. 39–67.
- Brunnermeier, Markus K. and Yuliy Sannikov (2014) "A Macroeconomic Model with a Financial Sector," *The American Economic Review*, Vol. 104, pp. 379–421.
- Buchak, Greg, Gregor Matvos, Tomasz Piskorski, and Amit Seru (2017) "Fintech, Regulatory Arbitrage, and the Rise of Shadow Banks," NBER Working Paper No. 23288.
- Carpinelli, Luisa and Matteo Crosignani (2017) "The Effect of Central Bank Liquidity Injections on Bank Credit Supply," FEDS Working Paper No. 2017-038.
- Chodorow-Reich, Gabriel (2014) "Effects of Unconventional Monetary Policy on Financial Institutions," *Brookings Papers on Economic Activity*, Vol. 2014, pp. 155–227.
- Cottarelli, Carlo, Giovanni Ferri, and Andrea Generale (1995) "Bank Lending Rates and Financial Structure in Italy: A Case Study," *IMF Staff Papers*, Vol. 42, pp. 670–700.
- Crawford, Gregory S., Nicola Pavanini, and Fabiano Schivardi (2018) "Asymmetric Information and Imperfect Competition in Lending Markets," *The American Economic Review* (Forthcoming).
- Crosignani, Matteo, Miguel Faria-e Castro, and Luis Fonseca (2017) "The (Unintended?) Consequences of the Largest Liquidity Injection Ever," Unpublished working paper.
- De Graeve, Ferre, Olivier De Jonghe, and Rudi Vander Vennet (2007) "Competition, Transmission and Bank Pricing Policies: Evidence from Belgian Loan and Deposit Markets," *Journal of Banking & Finance*, Vol. 31, pp. 259–278.
- De Marco, Filippo (2015) "Bank Lending and the European Sovereign Debt Crisis," Unpublished working paper.

- Di Maggio, Marco, Amir Kermani, and Christopher Palmer (2016) "How Quantitative Easing Works: Evidence on the Refinancing Channel," NBER Working Paper No. 22638.
- Drechsler, Itamar, Thomas Drechsel, David Marques-Ibanez, and Philipp Schnabl (2016) "Who Borrows From the Lender of Last Resort?" The Journal of Finance, Vol. 71, pp. 1933–1974.
- Drechsler, Itamar, Alexi Savov, and Philipp Schnabl (2017) "The Deposits Channel of Monetary Policy," *The Quarterly Journal of Economics*, Vol. 132, pp. 1819–1876.
- ECB (2014) "Targeted Longer-Term Refinancing Operations: Updated Guidelines for Completing the Reporting Template," Technical report, ECB.
- Felici, Roberto and Marcello Pagnini (2008) "Distance, Bank Heterogeneity and Entry in Local Banking Markets," The Journal of Industrial Economics, Vol. 56, pp. 500–534.
- Freixas, Xavier and Jean-Charles Rochet (2008) *Microeconomics of Banking*: MIT Press, 2nd edition.
- Gambacorta, Leonardo and Paolo Emilio Mistrulli (2014) "Bank Heterogeneity and Interest Rate Setting: What Lessons Have We Learned since Lehman Brothers?," Journal of Money, Credit and Banking, Vol. 46, pp. 753–778.
- Gerali, Andrea, Stefano Neri, Luca Sessa, and Federico M. Signoretti (2010) "Credit and Banking in a DSGE Model of the Euro Area," *Journal of Money, Credit and Banking*, Vol. 42, pp. 107–141.
- Gertler, Mark and Peter Karadi (2011) "A Model of Unconventional Monetary Policy," Journal of Monetary Economics, Vol. 58, pp. 17–34.
- Gertler, Mark and Nobuhiro Kiyotaki (2010) "Financial Intermediation and Credit Policy in Business Cycle Analysis," in Benjamin Friedman and Michael Woodford eds. *Handbook* of Monetary Economics, Vol. 3A: Elsevier, pp. 547–599.
- Guiso, Luigi, Paola Sapienza, and Luigi Zingales (2004) "Does Local Financial Development Matter?," The Quarterly Journal of Economics, Vol. 119, pp. 929–969.

Jiménez, Gabriel, Steven Ongena, José-Luis Peydró, and Jesús Saurina (2012) "Credit Sup-

ply and Monetary Policy: Identifying the Bank Balance-Sheet Channel with Loan Applications," *The American Economic Review*, Vol. 102, pp. 2301–2326.

- (2014) "Hazardous Times for Monetary Policy: What Do Twenty-Three Million Bank Loans Say About the Effects of Monetary Policy on Credit Risk-Taking?," *Econometrica*, Vol. 82, pp. 463–505.
- Kashyap, Anil K. and Jeremy C. Stein (2000) "What Do a Million Observations on Banks Say about the Transmission of Monetary Policy?," *The American Economic Review*, Vol. 90, pp. 407–428.
- Khwaja, Asim Ijaz and Atif Mian (2008) "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market," The American Economic Review, Vol. 98, pp. 1413– 1442.
- Krishnamurthy, Arvind, Stefan Nagel, and Annette Vissing-Jorgensen (2017) "ECB Policies Involving Government Bond Purchases: Impact and Channels," *Review of Finance*, Vol. 22, pp. 1–44.
- Langfield, Sam and Marco Pagano (2016) "Bank Bias in Europe: Effects on Systemic Risk and Growth," *Economic Policy*, Vol. 31, pp. 51–106.
- Mian, Atif (2006) "Distance Constraints: the Limits of Foreign Lending in Poor Economies," The Journal of Finance, Vol. 61, pp. 1465–1505.
- Neumark, David and Steven A. Sharpe (1992) "Market Structure and the Nature of Price Rigidity: Evidence from the Market for Consumer Deposits," *The Quarterly Journal of Economics*, Vol. 107, pp. 657–680.
- Pascali, Luigi (2016) "Banks and Development: Jewish Communities in the Italian Renaissance and Current Economic Performance," *Review of Economics and Statistics*, Vol. 98, pp. 140–158.
- Petersen, Mitchell A. and Raghuram G. Rajan (1994) "The Benefits of Lending Relation-ships: Evidence from Small Business Data," *The Journal of Finance*, Vol. 49, pp. 3–37.
  —— (1995) "The Effect of Credit Market Competition on Lending Relationships," *The Quarterly Journal of Economics*, Vol. 110, pp. 407–443.

- PwC (2010) "Cambiare per Sopravvivere: l'Evoluzione del Ruolo della Filiale in Italia," Technical report, PricewaterhouseCoopers.
- Rajan, Raghuram G. (1992) "Insiders and Outsiders: The Choice Between Informed and Arm's-length Debt," *The Journal of Finance*, Vol. 47, pp. 1367–1400.
- Rochet, Jean-Charles (2009) Why Are There So Many Banking Crises?: The Politics and Policy of Bank Regulation: Princeton University Press.
- Rodnyansky, Alexander and Olivier M. Darmouni (2017) "The Effects of Quantitative Easing on Bank Lending Behavior," *The Review of Financial Studies*, Vol. 30, pp. 3858–3887.
- Scharfstein, David and Adi Sunderam (2014) "Market Power in Mortgage Lending and the Transmission of Monetary Policy," Unpublished working paper.
- Schnabl, Philipp (2012) "The International Transmission of Bank Liquidity Shocks: Evidence from an Emerging Market," *The Journal of Finance*, Vol. 67, pp. 897–932.
- Sharpe, Steven A. (1990) "Asymmetric Information, Bank Lending, and Implicit Contracts: A Stylized Model of Customer Relationships," *The Journal of Finance*, Vol. 45, pp. 1069– 1087.
- Stackhouse, Julie (2018) "Why Are Banks Shuttering Branches?," Post written by Julie Stackhouse, executive vice president and officer-in-charge of supervision at the St. Louis Federal Reserve, as part of a series titled "Supervising Our Nation's Financial Institutions".
- Stock, James H. and Motohiro Yogo (2002) "Testing for Weak Instruments in Linear IV Regression," NBER Technical Working Paper No. 284.
- Van Leuvensteijn, Michiel, Christoffer Kok Sørensen, Jacob A. Bikker, and Adrian Van Rixtel (2013) "Impact of Bank Competition on the Interest Rate Pass-through in the Euro Area," *Applied Economics*, Vol. 45, pp. 1359–1380.
- Van Rixtel, Adrian and Gabriele Gasperini (2013) "Financial Crises and Bank Funding: Recent Experience in the Euro Area," BIS Working Paper No. 406.

# A Main Figures and Tables

## Figure 1: TLTRO timeline

The figure shows a timeline of the ECB TLTROS. On the 5th of June 2014, the ECB decided the first series of Targeted Longer-Term Refinancing Operations (TLTROS). The policy measure is implemented through eight auctions, one at the end of each quarter from the end of September 2014 to the end of June 2016. Banks borrowing in the first two TLTROS cannot exceeds 7% of the outstanding amount of eligible loans on 30 April 2014. All TLTROS will mature in September 2018, but banks have the option to repay any part of the amounts they were allotted in a TLTRO after 24 months at a biannual frequency. The ECB imposes a mandatory early repayment in September 2016, if some lending requirements are not satisfied. A second series of four operations starting in June 2016 has been announced on the 10th of March 2016.



## Figure 2: Geographical distribution of Herfindahl Index

The figure shows the geographical distribution of the quartiles of the Herfindahl index in the term loan sector. The index is calculated using quantities of credit for each province in the first quarter of 2014. We used the structure of 103 Italian provinces existing until 2005 to get a homogeneous classification of the provinces from the different datasets. The credit market in Italy is relatively concentrated, with an average value of the index of 0.17 and a range of values included between 0.09 and 0.36 (see Table 1).



## Figure 3: Geographical distribution of pawnshops

The figure shows the geographical distribution of the number of pawnshop during the Renaissance. The index is calculated aggregating the number of pawnshops by cities using the structure of 103 Italian provinces existing until 2005. The number of pawnshops comes from Pascali (2016).



## Table 1: Descriptive statistics

The table shows the summary statistics for the main variables in our analysis. Panel A shows the overall interest rate, with and without accessory expenses, on the stock of term loans and including expenditure for the flows of new loans. In Panel B, Herfindahl index at province level is calculated using quantities of credit for each province; the one at province and rating level is constructed segmenting markets by province, separately for firms of average and high credit quality (classes 1-6 of the rating index) and firms of low quality (classes 7-9). Pawnshops is the total number of pawnshops in each province using data from Pascali (2016). In Panel C amount borrowed is the total amount borrowed in the first two TLTROs; additional amount borrowed is the corrected amount of the exposition to central bank coming from the first two TLTROS, netting out the debts towards the Eurosystem expiring in the same quarter; maximum allowance is the borrowing limit computed from expression (1). In panel D we report the main structural characteristics of the banks in the first quarter of 2014: total assets and ratio of government bonds, total loans, bad loans and capital. In Panel E we report the distribution of firm assets and compare the statistics where the statistical unit is the firm with those where the statistical unit is the firm-bank relationship (which is the one we use in our final dataset); the statistics are substantially similar in both cases, taking into account that on average bigger firms have more credit relationships and therefore their weight is bigger when the statistical unit is the relationship.

	Obs	Mean	Std. Dev.	Min	Median	Max
PANEL A: TRANSACTION LEVEL VARIA	BLES $(1s)$	T QUAR	TER 2014-2	ND QU	ARTER 20	15)
Interest rate incl. expenditures $(\%)$	671951	4.06	1.96	0.49	3.85	12.10
Interest rate w/out expenditures $(\%)$	671951	4.05	1.95	0.49	3.85	11.83
Interest rate incl. expenditures (flows; $\%$ )	58098	4.85	2.30	0.50	4.56	14.31
PANEL B: PROVINCE LEVEL V	ARIABLES	6 (1ST G	QUARTER 20	14 for	HI)	
Province level HI on credit amount	103	0.17	0.06	0.09	0.16	0.36
Province - Rating 1-6 HI	103	0.17	0.06	0.08	0.16	0.36
Province - Rating 7-9 HI	103	0.18	0.08	0.09	0.16	0.48
Pawnshops (number)	103	1.02	0.50	0.00	0.0	8.00
Panel C: I-II TLT	RO VARI	ABLES (	BANK LEVE	L)		
Amount borrowed (million euros)	78	670.0	1843	5	85.72	12500
Additional amount borrowed (million euros)	43	542.7	1167	5	123	5495
Maximum allowance (million euros)	104	560.3	1635	16.11	83.49	12500
Panel D: Other bank le	VEL VARI	ABLES (	(1st quart	er 201	4)	
Assets (billion euros)	104	30.19	101.27	0.46	2.97	777.91
Loans over assets ratio $(\%)$	104	54.07	11.64	8.21	55.75	74.86
Bad loans over loans ratio $(\%)$	104	9.33	5.44	0.09	8.75	27.57
Government bonds over assets ratio $(\%)$	104	18.26	8.86	1.09	18.25	43.30
Capital ratio (%)	104	15.48	9.24	0.25	13.86	94.89
PANEL E: FIRM	LEVEL V	ARIABLI	ES(2013)			
Assets (million euros; by firm)	73174	3.95	30.08	1	0.59	2548.20
Assets (million euros; by relationship)	113246	7.24	40.66	1	0.95	2548.20
		]	Percentage d	istribut	ion	
Classes:		1-6			7-9	
Credit rating (by firm)		73%			27%	
Credit rating (by relationship)		74%			26%	

	Table 2:	Descriptive	statistics	for	treated	and	controls
--	----------	-------------	------------	-----	---------	-----	----------

The table shows the summary statistics for the main variables in our analysis in the group of treated and control banks. Panel A shows the main the overall interest rate, with and without accessory expenses, on the stock of term loans and including expenditure for the flows of new loans. In Panel B maximum allowance is the borrowing limit computed from expression (1). In panel C we report the main structural characteristics of the banks in the first quarter of 2014: total assets and ratio of government bonds, total loans, bad loans and capital. In Panel D we report the distribution of firm assets and the percentages of firms whose credit quality is either high/average (about three quarters of the sample) or low, using as statistical unit the firmbank relationship (which is the relevant statistical unit in our final dataset).

		Treate	ed		Contro	ls
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.
PANEL A: TRANSACTION L	EVEL VAF	RIABLES	(1st-3rd Qu)	JARTER 2	014)	
Interest rate incl. expenditures $(\%)$	220776	4.23	2.04	115046	4.14	1.84
Interest rate w/out expenditures $(\%)$	220776	4.22	2.03	115046	4.12	1.82
Interest rate incl. expenditures (flows; %)	21243	5.21	2.27	7806	4.74	2.16
Panel B: I-II T	LTRO V.	ARIABLE	es (bank lev	'EL)		
Max allowance (million euros)	43	841.7	2238	61	359	979.1
PANEL C: OTHER BANK	LEVEL V	ARIABLI	ES $(1$ ST QUAR	RTER $2014$	1)	
Assets (billion euros)	43	47.87	145.61	61	17.72	48.81
Loans over assets ratio $(\%)$	43	53.20	10.25	61	54.68	12.58
Bad loans over loans ratio $(\%)$	43	7.85	2.65	61	10.38	6.57
Government bonds over assets ratio $(\%)$	43	17.93	7.79	61	18.50	9.60
Capital ratio (%)	43	16.75	12.68	61	14.58	5.64
Panel D: Fi	RM LEVE	L VARIA	BLES $(2013)$			
Assets (million euros)	74372	6.75	38.14	38874	8.17	45.04
			Percentage	distributio	n	
		Т	reated		$\mathbf{C}$	ontrols
Classes:		1-6	7-9		1-6	7-9
Credit rating		75%	25%		72%	28%

#### Table 3: Targeted monetary policy - OLS estimates

The table reports the estimated parameters and their standard errors from the OLS estimation of equation (2). Column (1) reports the estimates with the full balanced dataset. Columns (2) and (3) report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variable is the interest rate including expenditure on the stock of loan from bank b to firm f in quarter t. Binary treatment is a dummy equal to one if the bank borrows from the TLTROS. Continuous treatment is a continuous variable equal to the logarithm of the actual additional amount the bank borrows from the TLTROS. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Binary treatment			Continuous treatment			
	All	Wit	thin	All	Wi	thin	
	(1)	(2)	(3)	(4)	(5)	(6)	
TLTROs $\times$							
2014 - Q3	-0.034	-0.025	-0.032	-0.0019	-0.0014	-0.0017	
	(0.041)	(0.028)	(0.028)	(0.002)	(0.0015)	(0.0014)	
2014 - Q4	0.025	0.011	-0.013	0.001	0.00047	-0.00061	
	(0.035)	(0.022)	(0.023)	(0.0017)	(0.0011)	(0.0011)	
2015 - Q1	-0.011	-0.024	$-0.047^{*}$	-0.0011	-0.0015	-0.0026**	
	(0.039)	(0.024)	(0.025)	(0.0018)	(0.0012)	(0.0012)	
2015 - Q2	-0.019	$-0.051^{**}$	-0.065**	-0.0014	$-0.0027^{**}$	-0.0033***	
	(0.047)	(0.024)	(0.025)	(0.0023)	(0.0012)	(0.0013)	
Firm f.e.	Yes	No	No	Yes	No	No	
Time f.e.	Yes	No	No	Yes	No	No	
Bank f.e.	Yes	Yes	Yes	Yes	Yes	Yes	
Firm-time f.e.	No	Yes	Yes	No	Yes	Yes	
Bank-time controls	No	No	Yes	No	No	Yes	
Observations	654,948	354,600	354,060	654,948	354,600	$354,\!060$	
Adjusted $R^2$	0.71	0.36	0.36	0.71	0.36	0.36	

## Table 4: Targeted monetary policy - IV estimates

The table reports the estimated parameters and their standard errors from the IV estimation of equation (4). Column (1) reports the estimates with the full balanced dataset. Columns (2) and (3) report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variable is the interest rate including expenditure on the stock of loan from bank b to firm f in quarter t. Binary treatment is a dummy equal to one if the bank borrows from the TLTROS. Continuous treatment is a continuous variable equal to the logarithm of the actual additional amount the bank borrows from the TLTROS. Both the binary and the continuous treatment are instrumented. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Bin	Binary treatment			Continuous treatment			
	All	Wi	thin	All	Within			
	(1)	(2)	(3)	(4)	(5)	(6)		
TLTROs $\times$								
2014 - Q3	-0.18	-0.12	-0.12	-0.0067	-0.0043	-0.0044		
	(0.14)	(0.097)	(0.082)	(0.005)	(0.0035)	(0.003)		
2014 - Q4	-0.049	0.039	0.076	-0.0018	0.0014	0.0026		
	(0.12)	(0.083)	(0.069)	(0.004)	(0.0028)	(0.0023)		
2015 - Q1	-0.39**	-0.23**	-0.20***	-0.014**	-0.0085**	-0.0073***		
	(0.18)	(0.098)	(0.076)	(0.0057)	(0.0033)	(0.0026)		
2015 - Q2	-0.36*	-0.24*	$-0.19^{*}$	-0.013*	-0.0087**	$-0.0071^{*}$		
	(0.21)	(0.13)	(0.11)	(0.0071)	(0.0044)	(0.0041)		
Firm f.e.	Yes	No	No	Yes	No	No		
Time f.e.	Yes	No	No	Yes	No	No		
Bank f.e.	Yes	Yes	Yes	Yes	Yes	Yes		
Firm-time f.e.	No	Yes	Yes	No	Yes	Yes		
Bank-time controls	No	No	Yes	No	No	Yes		
Observations	654 948	354 600	354 060	654 948	354 600	354 060		
Adjusted $R^2$	0.71	0.36	0.36	0.71	0.36	0.36		

#### Table 5: Targeted monetary policy and competition

The table reports the estimated parameters and their standard errors from the IV estimation of equations (5) and (7). All columns report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variable is the interest rate including expenditure on the stock of loan from bank b to firm f in quarter t. Binary treatment is a dummy equal to one if the bank borrows from the TLTROS. Continuous treatment is a continuous variable equal to the logarithm of the actual additional amount the bank borrows from the TLTROS. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Binary tr	eatment	Continuous	s treatment
	(IV-OLS)	(IV-IV)	(IV-OLS)	(IV-IV)
	(1)	(2)	(3)	(4)
TLTROs $\times$				
2014 - Q3	-0.48**	-1.60**	-0.018**	-0.061**
	(0.23)	(0.69)	(0.0088)	(0.025)
2014 - Q4	-0.44*	-1.38**	-0.016*	-0.054**
	(0.25)	(0.69)	(0.009)	(0.025)
2015 - Q1	-0.38*	-2.08***	-0.014*	-0.077***
	(0.24)	(0.8)	(0.0086)	(0.028)
2015 - Q2	-0.42*	-1.71**	-0.016*	-0.063**
	(0.24)	(0.76)	(0.0086)	(0.027)
TLTROs $\times$ HI $\times$		. ,	. ,	. ,
2014 - Q3	$2.61^{*}$	$10.8^{**}$	$0.097^{*}$	$0.41^{**}$
	(1.56)	(4.87)	(0.058)	(0.18)
2014 - Q4	$3.72^{*}$	$10.4^{**}$	0.13**	$0.40^{**}$
	(1.91)	(4.97)	(0.065)	(0.18)
2015 - Q1	1.29	$13.6^{**}$	0.052	$0.50^{**}$
	(1.81)	(5.68)	(0.064)	(0.20)
2015 - Q2	1.66	$10.9^{**}$	0.065	$0.40^{**}$
	(2.02)	(5.46)	(0.071)	(0.19)
Firm-time f.e.	Yes	Yes	Yes	Yes
Bank f.e.	Yes	Yes	Yes	Yes
Bank-time controls	Yes	Yes	Yes	Yes
Observations	354,060	354,060	354,060	354,060
Adjusted $\mathbb{R}^2$	0.36	0.35	0.36	0.35

#### Table 6: Targeted monetary policy and competition - Heterogeneity

The table reports the estimated parameters and their standard errors from the IV estimation of equation (5) in different subsets of the data. All columns report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variable is the interest rate including expenditure on the stock of loan from bank b to firm f in quarter t. TLTROs is a dummy equal to one if the bank borrows from the TLTROs. High risk firms are firms with a bad credit score (7-9), while low risk are firms with a good or average credit rating (classes 1-6). Small firms are firms below the median of the distribution of assets in the pre-treatment year (2013). Large banks are the top 5 banks in Italy. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Firm	n risk	Firm size		Bank	size
	High	Low	Small	Large	Large	Small
	(1)	(2)	(3)	(4)	(5)	(6)
TLTROs $\times$						
2014 - Q3	-0.59	-0.34**	-0.65**	-0.37	-0.43**	0.46
	(0.48)	(0.19)	(0.33)	(0.27)	(0.21)	(0.80)
2014 - Q4	-0.25	-0.28	-0.58	-0.32	-0.55***	1.30
	(0.54)	(0.19)	(0.39)	(0.27)	(0.19)	(1.02)
2015 - Q1	-0.58	-0.20	-0.53	-0.30	-0.55***	1.18
	(0.50)	(0.18)	(0.37)	(0.26)	(0.17)	(1.34)
2015 - Q2	-0.18	-0.43**	-0.78**	-0.22	-0.36	0.05
	(0.47)	(0.18)	(0.36)	(0.28)	(0.23)	(1.40)
TLTROs $\times$ HI $\times$	. ,		· /	· /	( )	· · ·
2014 - Q3	4.35	1.36	3.29	2.18	1.90	-3.56
	(3.77)	(1.22)	(2.37)	(1.87)	(1.22)	(6.85)
2014 - Q4	2.84	$2.36^{**}$	$4.95^{*}$	2.76	3.71***	-9.61
	(3.89)	(1.35)	(2.90)	(2.00)	(1.26)	(9.85)
2015 - Q1	3.04	-0.03	1.72	1.11	3.20***	9.81
	(3.47)	(1.27)	(2.62)	(2.07)	(1.12)	(12.30)
2015 - Q2	0.65	1.43	3.61	0.60	$2.64^{*}$	1.14
	(3.16)	(1.32)	(2.60)	(2.43)	(1.33)	(11.90)
Firm-time f.e.	Yes	Yes	Yes	Yes	Yes	Yes
Bank f.e.	Yes	Yes	Yes	Yes	Yes	Yes
Bank-time controls	Yes	Yes	Yes	Yes	Yes	Yes
Observations	81.930	272,130	154,458	199,602	135.936	106.680
Adjusted $\mathbb{R}^2$	0.30	0.36	0.31	0.34	0.41	0.34

# **B** Additional Figures and Tables

#### Table 7: Targeted monetary policy - First stage

The table reports the estimated parameters and their standard errors for the first stage of the IV model of equation (4). All columns report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variables are the TLTROs binary and continuous treatments. Rule is the allocation rule for bank *b* from equation (1), Post is a dummy equal to one after the implementation of the TLTROs. Bank-quarter control includes bank capital, non-performing loans, government bonds. The Kleibergen-Paap F-statistic test for weak instruments with cluster-robust standard errors. The Kleibergen-Paap LM-statistic test for underidentification with cluster-robust standard errors. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Binary treatment	Continuous treatment
	(1)	(2)
$Rule \times Post$	2.15***	0.080***
	(0.287)	(0.012)
Firm-time f.e.	Yes	Yes
Bank f.e.	Yes	Yes
Bank-time controls	Yes	Yes
Kleibergen-Paap F-statistic	41.81	56.36
Kleibergen-Paap LM-statistic	17.44	20.52
Observations	354,060	354,060
Adjusted $\mathbb{R}^2$	0.82	0.82

## Table 8: Targeted monetary policy and competition - First stage

The table reports the estimated parameters and their standard errors for the first stage of the IV model of equation (7). All columns report estimates with the balanced panel of relationships for firms with more than one lender. The dependent variables in columns (1) and (3) are the TLTROs binary and continuous treatments, while in columns (2) and (4) the dependent variables are the interactions with the HI. Rule is the allocation rule for bank b from equation (1). Post is a dummy equal to one after the implementation of the TLTROs. Pawnshop is the number of pawnshops across Italian provinces during the Renaissance. Bank-time controls include bank capital, non-performing loans, government bonds. The Kleibergen-Paap F-statistic test for weak instruments with cluster-robust standard errors. The Kleibergen-Paap LM-statistic test for underidentification with cluster-robust standard errors. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

Binary t	Binary treatment		s treatment
(1)	(2)	(3)	(4)
1.825***	-0.548	0.073***	-0.001
(0.278)	(0.058)	(0.011)	(0.003)
2.428*	$2.645^{***}$	0.051	0.091***
(1.313)	(0.461)	(0.054)	(0.021)
Yes	Yes	Yes	Yes
Yes	Yes	Yes	Yes
Yes	Yes	Yes	Yes
10.49	10.02	15.57	13.86
9.81	10.46	12.46	12.87
354,060	354,060	354,060	354,060
0.82	0.81	0.82	0.81
	$\begin{array}{r} {\rm Binary \ t} \\ \hline (1) \\ 1.825^{***} \\ (0.278) \\ 2.428^{*} \\ (1.313) \\ {\rm Yes} \\ 10.49 \\ 9.81 \\ 354,060 \\ 0.82 \end{array}$	$\begin{tabular}{ c c c c } \hline Binary treatment \\\hline (1) & (2) \\\hline 1.825^{***} & -0.548 \\(0.278) & (0.058) \\2.428^* & 2.645^{***} \\(1.313) & (0.461) \\\hline Yes & Yes \\Yes & Yes \\Yes & Yes \\Yes & Yes \\10.49 & 10.02 \\9.81 & 10.46 \\354,060 & 354,060 \\0.82 & 0.81 \\\hline \end{tabular}$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

## Table 9: Targeted monetary policy and competition - Main - Robustness

The table reports the estimated parameters and their standard errors from the IV estimation of equation (5) for different robustness exercises. All columns report estimates with the balanced panel of firms with more than one lender. The dependent variable is the interest rate on the loan from bank b to firm f in quarter t. Binary treatment is a dummy equal to one if the bank borrows from the TLTROS. Continuous treatment is a continuous variable equal to the logarithm of the actual additional amount the bank borrows from the TLTROS. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	No c	orrection	No ex	penditure
	Binary (1)	Continuous (2)	Binary (3)	Continuous (4)
TLTROs $\times$				
2014 - Q3	-0.92	-0.024*	-0.53**	-0.020**
	(0.56)	(0.014)	(0.23)	(0.0086)
2014 - Q4	-0.78	-0.021	-0.52**	-0.019**
	(0.54)	(0.014)	(0.25)	(0.0087)
2015 - Q1	-0.85*	-0.021*	-0.46**	-0.017**
	(0.51)	(0.013)	(0.23)	(0.0082)
2015 - Q2	-0.96*	-0.024*	-0.51**	-0.019**
	(0.54)	(0.013)	(0.23)	(0.0083)
TLTROs $\times$ HI $\times$	· · · ·	× ,	· /	
2014 - Q3	4.91	0.13	$3.01^{**}$	$0.11^{**}$
	(3.57)	(0.091)	(1.53)	(0.056)
2014 - Q4	6.37	$0.17^{*}$	4.22**	0.15**
	(3.95)	(0.099)	(1.91)	(0.063)
2015 - Q1	3.17	0.081	1.78	0.07
	(3.54)	(0.09)	(1.71)	(0.061)
2015 - Q2	3.84	0.099	2.26	0.087
	(3.92)	(0.099)	(1.88)	(0.067)
Firm-time f.e.	Yes	Yes	Yes	Yes
Bank f.e.	Yes	Yes	Yes	Yes
Bank-time controls	Yes	Yes	Yes	Yes
Observations	354,060	354,060	354,060	354,060
Adjusted $\mathbb{R}^2$	0.36	0.36	0.36	0.36

Table 10: Targeted monetary policy and competition - New loans - Robustness The table reports the estimated parameters and their standard errors from the IV estimation of equations (4) and (5) for the flows of new loans. All columns report estimates with the balanced panel of firms with more than one lender. The dependent variable is the interest rate on the loan from bank b to firm f in quarter t. Binary treatment is a dummy equal to one if the bank borrows from the TLTROS. Continuous treatment is a continuous variable equal to the logarithm of the actual additional amount the bank borrows from the TLTROS. Bank-time controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Direct	effect only	Includin with c	g interaction competition
	Binary (1)	Continuous (2)	Binary (3)	Continuous (4)
TLTROs $\times$				
2014 - Q3	-0.12	-0.004	-1.92	-0.054
	(0.15)	(0.004)	(1.78)	(0.034)
2014 - Q4	-0.34**	-0.014***	-1.77	-0.055*
	(0.15)	(0.003)	(1.38)	(0.028)
2015 - Q1	-0.73***	-0.025***	-2.42	-0.073*
	(0.15)	(0.004)	(1.64)	(0.032)
2015 - Q2	-0.91***	-0.031***	$-2.31^{*}$	-0.071*
	(0.23)	(0.005)	(1.33)	(0.025)
TLTROs $\times$ HI $\times$				
2014 - Q3			8.17	-0.23
			(8.32)	(0.16)
2014 - Q4			6.51	-0.18
			(6.22)	(0.12)
2015 - Q1			7.71	0.22
			(7.15)	(0.14)
2015 - Q2			6.4	$0.18^{*}$
			(5.56)	(0.11)
Firm-time f.e.	Yes	Yes	Yes	Yes
Bank f.e.	Yes	Yes	Yes	Yes
Bank-time controls	Yes	Yes	Yes	Yes
Observations	33,528	$33,\!528$	$33,\!528$	$33,\!528$
Adjusted $\mathbb{R}^2$	0.68	0.69	0.68	0.69

## Table 11: Targeted monetary policy and competition - Heterogeneity - Robustness

The table reports the estimated parameters and their standard errors from the IV estimation of equation (5) in different subsets of the data. All columns report estimates with the balanced panel of relationship for firms with more than one lender. The dependent variable is the interest rate including expenditure on the stock of loan from bank b to firm f in quarter t. TLTROs is the logarithm of the actual additional amount the bank borrows from the TLTROs. High risk firms are firms with a bad credit score (7-9), while low risk are firms with a good or average credit rating (classes 1-6). Small firms are firms below the median of the distribution of assets in the pre-treatment year (2013). Large banks are the top 5 banks in Italy. Bank-quarter controls include bank capital, non-performing loans, government bonds. All standard errors are double clustered by firm and bank-quarter. \*,\*\*, \*\*\* indicates significance at the 0.10, 0.05 and 0.01 levels, respectively.

	Firm risk		Firm size		Bank size	
	High	Low	Small	Large	Large	Small
	(1)	(2)	(3)	(4)	(5)	(6)
TLTROs $\times$						
2014 - Q3	-0.018	-0.013*	-0.024**	-0.014	-0.018*	0.034
	(0.013)	(0.007)	(0.012)	(0.01)	(0.008)	(0.071)
2014 - Q4	-0.006	-0.011	-0.02	-0.012	-0.023***	0.094
	(0.015)	(0.007)	(0.013)	(0.01)	(0.008)	(0.12)
2015 - Q1	-0.02	-0.008	-0.02	-0.012	-0.023***	-0.09
	(0.014)	(0.006)	(0.013)	(0.009)	(0.007)	(0.16)
2015 - Q2	-0.006	-0.017**	-0.029**	-0.009	-0.014	-0.011
	(0.015)	(0.006)	(0.013)	(0.01)	(0.009)	(0.12)
TLTROs $\times$ HI $\times$						
2014 - Q3	0.13	0.054	0.12	0.082	0.08	-0.28
	(0.099)	(0.047)	(0.083)	(0.07)	(0.05)	(0.62)
2014 - Q4	0.078	$0.088^{*}$	$0.17^{*}$	0.098	$0.16^{***}$	-0.73
	(0.10)	(0.049)	(0.094)	(0.074)	(0.054)	(1.08)
2015 - Q1	0.097	0.001	0.067	0.045	$0.14^{***}$	0.75
	(0.10)	(0.048)	(0.091)	(0.075)	(0.048)	(1.42)
2015 - Q2	0.024	0.058	0.13	0.026	$0.11^{*}$	0.15
	(0.094)	(0.05)	(0.091)	(0.087)	(0.058)	(1.03)
Firm-time f.e	Ves	Ves	Ves	Ves	Ves	Ves
Bank fe	Ves	Ves	Ves	Ves	Ves	Ves
Bank-time controls	Ves	Ves	Ves	Ves	Ves	Ves
Dank-time controls	105	105	105	105	105	105
Observations	81,930	272,130	$154,\!458$	199,602	$135,\!936$	106,680
Adjusted $\mathbb{R}^2$	0.30	0.36	0.31	0.34	0.41	0.33

# C Additional Features of TLTROs

In this appendix we briefly describe some additional features of the TLTROS. The borrowing limit on the third to eight TLTROS is differently computed from the first two operations. The ECB defines a benchmark  $BE_b^k$  for each bank b in each TLTRO k given by the formula:

$$BE_b^k = 0 \qquad for \ k = 3, ..., 8 \quad if \ \overline{NL} \ge 0$$
  
$$BE_b^k = \overline{NL} \times n_k \quad for \ k = 3, ..., 8 \quad if \ \overline{NL} < 0$$
(8)

where  $\overline{NL} = \frac{(NL_b^{May2013} + ... + NL_b^{Apr2014})}{12}$  is the average eligible net lending of institution b from May 2013 to April 2014 and  $n_k = 9$  for k = 3 and  $n_k = 12$  for k = 4, ..., 8.<sup>19</sup> The additional borrowing limit is then computed as:

$$q_b^k \leq 3(CNL_b^k - BE_b^k) - \sum_{j=3}^{k-1} q_b^j \text{ for } k = 3, ..., 8$$
 (9)

where  $CNL_b^k = NL_b^{May2014} + ... + NL_b^{Month(k)-2}$  is the cumulative net lending in operations from May 2014 until two months before operations k takes place.

Finally, the ECB set also some special rules for the TLTROS on repayment. Even if all TLTROS will mature in September 2018, there are prepayment options and a mandatory repayment rule. On the one hand, intermediaries have the option to repay any part of the amounts they were allotted in a TLTRO after 24 months at a biannual frequency. On the other hand, the ECB imposes a mandatory early repayment  $(MR_b)$  in September 2016, if some lending requirements are not satisfied. The early repayment rule is applied according to the following formula:

$$MR_{b} = \sum_{j=1}^{8} q_{b}^{j} \qquad if \ BE_{b}^{8} > CNL_{b}^{8}$$

$$MR_{b} = \sum_{j=3}^{8} q_{b}^{j} - 3(CNL_{b}^{8} - BE_{b}^{8}) \quad if \ BE_{b}^{8} \le CNL_{b}^{8}.$$
(10)

<sup>&</sup>lt;sup>19</sup> "Eligible net lending" means gross lending in the form of eligible loans net of repayments of outstanding amounts of eligible loans during a specific period. For details see again ECB (2014).

Thus the bank has to repay the whole borrowed amount through the TLTROs if the total eligible net lending in the period May 2014-April 2016  $(CNL_b^8)$  is less than the benchmark for the last operation. Otherwise, the bank has to pay back in September 2016 the amount borrowed in the last six TLTROs in excess of the amount used for the calculation of the additional allowance for the last operations, that is thrice the cumulative net lending exceeding the benchmark.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 1163 What will Brexit mean for the British and euro-area economies? A model-based assessment of trade regimes, by Massimiliano Pisani and Filippo Vergara Caffarelli (January 2018).
- N.1164 Are lenders using risk-based pricing in the consumer loan market? The effects of the 2008 crisis, by Silvia Magri (January 2018).
- N.1165 Listening to the buzz: social media sentiment and retail depositors' trust by Matteo Accornero and Mirko Moscatelli (February 2018)
- N.1166 *Banks' holdings of and trading in government bonds*, by Michele Manna and Stefano Nobili (March 2018).
- N.1167 Firms' and households' investment in Italy: the role of credit constraints and other macro factors, by Claire Giordano, Marco Marinucci and Andrea Silvestrini (March 2018).
- N.1168 *Credit supply and productivity growth*, by Francesco Manaresi and Nicola Pierri (March 2018).
- N. 1169 *Consumption volatility risk and the inversion of the yield curve*, by Adriana Grasso and Filippo Natoli (March 2018).
- N.1170 *The global component of inflation volatility*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (April 2018).
- N. 1171 *The potential of big housing data: an application to the Italian real-estate market*, by Michele Loberto, Andrea Luciani and Marco Pangallo (April 2018).
- N. 1172 ECB monetary policy and the euro exchange rate, by Martina Cecioni (April 2018).
- N.1173 *Firms' investments during two crises*, by Antonio De Socio and Enrico Sette (April 2018).
- N.1174 How can the government spending multiplier be small at the zero lower bound?, by Valerio Ercolani and João Valle e Azevedo (April 2018).
- N. 1175 Asset price volatility in EU-6 economies: how large is the role played by the ECB?, by Alessio Ciarlone and Andrea Colabella (June 2018).
- N.1176 Fixed rate versus adjustable rate mortgages: evidence from euro area banks, by Ugo Albertazzi, Fulvia Fringuellotti and Steven Ongena (June 2018).
- N. 1177 Short term forecasts of economic activity: are fortnightly factors useful?, by Libero Monteforte and Valentina Raponi (June 2018).
- N. 1178 *Discretion and supplier selection in public procurement*, by Audinga Baltrunaite, Cristina Giorgiantonio, Sauro Mocetti and Tommaso Orlando (June 2018).
- N. 1179 *Labor market and financial shocks: a time varying analysis*, by Francesco Corsello and Valerio Nispi Landi (June 2018).
- N. 1180 On the unintended effects of public transfers: evidence from EU funding to Southern Italy, by Ilaria De Angelis, Guido de Blasio and Lucia Rizzica (June 2018).
- N. 1181 Always look on the bright side? Central counterparties and interbank markets during the financial crisis, by Massimiliano Affinito and Matteo Piazza (July 2018).
- N. 1182 *Knocking on parents' doors: regulation and intergenerational mobility*, by Sauro Mocetti, Giacomo Roma and Enrico Rubolino (July 2018).
- N. 1183 Why do banks securitise their assets? Bank-level evidence from over one hundred countries in the pre-crisis period, by Fabio Panetta and Alberto Franco Pozzolo.
- N. 1184 Capital controls spillovers, by Valerio Nispi Landi (July 2018).
- N.1185 The macroeconomic effects of an open-ended asset purchase programme, by Lorenzo Burlon, Alessandro Notarpietro and Massimiliano Pisani (July 2018).
- N.1186 Fiscal buffers, private debt and recession: the good, the bad and the ugly, by Nicoletta Batini, Giovanni Melina and Stefania Villa (July 2018).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- ALBANESE G., G. DE BLASIO and P. SESTITO, *My parents taught me. evidence on the family transmission of values,* Journal of Population Economics, v. 29, 2, pp. 571-592, **TD No. 955 (March 2014).**
- ANDINI M. and G. DE BLASIO, Local development that money cannot buy: Italy's Contratti di Programma, Journal of Economic Geography, v. 16, 2, pp. 365-393, **TD No. 915 (June 2013).**
- BARONE G. and S. MOCETTI, *Inequality and trust: new evidence from panel data*, Economic Inquiry, v. 54, pp. 794-809, **TD No. 973 (October 2014).**
- BELTRATTI A., B. BORTOLOTTI and M. CACCAVAIO, Stock market efficiency in China: evidence from the split-share reform, Quarterly Review of Economics and Finance, v. 60, pp. 125-137, TD No. 969 (October 2014).
- BOLATTO S. and M. SBRACIA, *Deconstructing the gains from trade: selection of industries vs reallocation of workers*, Review of International Economics, v. 24, 2, pp. 344-363, **TD No. 1037 (November 2015).**
- BOLTON P., X. FREIXAS, L. GAMBACORTA and P. E. MISTRULLI, *Relationship and transaction lending in a crisis*, Review of Financial Studies, v. 29, 10, pp. 2643-2676, **TD No. 917 (July 2013).**
- BONACCORSI DI PATTI E. and E. SETTE, Did the securitization market freeze affect bank lending during the financial crisis? Evidence from a credit register, Journal of Financial Intermediation, v. 25, 1, pp. 54-76, TD No. 848 (February 2012).
- BORIN A. and M. MANCINI, Foreign direct investment and firm performance: an empirical analysis of *Italian firms*, Review of World Economics, v. 152, 4, pp. 705-732, **TD No. 1011 (June 2015).**
- BRAGOLI D., M. RIGON and F. ZANETTI, *Optimal inflation weights in the euro area*, International Journal of Central Banking, v. 12, 2, pp. 357-383, **TD No. 1045 (January 2016).**
- BRANDOLINI A. and E. VIVIANO, *Behind and beyond the (headcount) employment rate,* Journal of the Royal Statistical Society: Series A, v. 179, 3, pp. 657-681, **TD No. 965 (July 2015).**
- BRIPI F., *The role of regulation on entry: evidence from the Italian provinces*, World Bank Economic Review, v. 30, 2, pp. 383-411, **TD No. 932 (September 2013).**
- BRONZINI R. and P. PISELLI, *The impact of R&D subsidies on firm innovation*, Research Policy, v. 45, 2, pp. 442-457, **TD No. 960 (April 2014).**
- BURLON L. and M. VILALTA-BUFI, A new look at technical progress and early retirement, IZA Journal of Labor Policy, v. 5, **TD No. 963 (June 2014).**
- BUSETTI F. and M. CAIVANO, *The trend-cycle decomposition of output and the Phillips Curve: bayesian estimates for Italy and the Euro Area,* Empirical Economics, V. 50, 4, pp. 1565-1587, **TD No. 941** (November 2013).
- CAIVANO M. and A. HARVEY, *Time-series models with an EGB2 conditional distribution*, Journal of Time Series Analysis, v. 35, 6, pp. 558-571, **TD No. 947 (January 2014).**
- CALZA A. and A. ZAGHINI, *Shoe-leather costs in the euro area and the foreign demand for euro banknotes,* International Journal of Central Banking, v. 12, 1, pp. 231-246, **TD No. 1039 (December 2015).**
- CESARONI T. and R. DE SANTIS, *Current account "core-periphery dualism" in the EMU*, The World Economy, v. 39, 10, pp. 1514-1538, **TD No. 996 (December 2014).**
- CIANI E., *Retirement, Pension eligibility and home production,* Labour Economics, v. 38, pp. 106-120, **TD** No. 1056 (March 2016).
- CIARLONE A. and V. MICELI, Escaping financial crises? Macro evidence from sovereign wealth funds' investment behaviour, Emerging Markets Review, v. 27, 2, pp. 169-196, TD No. 972 (October 2014).
- CORNELI F. and E. TARANTINO, *Sovereign debt and reserves with liquidity and productivity crises*, Journal of International Money and Finance, v. 65, pp. 166-194, **TD No. 1012 (June 2015).**
- D'AURIZIO L. and D. DEPALO, An evaluation of the policies on repayment of government's trade debt in *Italy*, Italian Economic Journal, v. 2, 2, pp. 167-196, **TD No. 1061 (April 2016).**
- DE BLASIO G., G. MAGIO and C. MENON, Down and out in Italian towns: measuring the impact of economic downturns on crime, Economics Letters, 146, pp. 99-102, TD No. 925 (July 2013).
- DOTTORI D. and M. MANNA, *Strategy and tactics in public debt management*, Journal of Policy Modeling, v. 38, 1, pp. 1-25, **TD No. 1005 (March 2015).**

- LIBERATI D., M. MARINUCCI and G. M. TANZI, Science and technology parks in Italy: main features and analysis of their effects on hosted firms, Journal of Technology Transfer, v. 41, 4, pp. 694-729, TD No. 983 (November 2014).
- MARCELLINO M., M. PORQUEDDU and F. VENDITTI, Short-Term GDP forecasting with a mixed frequency dynamic factor model with stochastic volatility, Journal of Business & Economic Statistics, v. 34, 1, pp. 118-127, TD No. 896 (January 2013).
- RODANO G., N. SERRANO-VELARDE and E. TARANTINO, *Bankruptcy law and bank financing*, Journal of Financial Economics, v. 120, 2, pp. 363-382, **TD No. 1013 (June 2015).**
- ZINNA G., Price pressures on UK real rates: an empirical investigation, Review of Finance, v. 20, 4, pp. 1587-1630, **TD No. 968 (July 2014).**

2017

- ADAMOPOULOU A. and G.M. TANZI, Academic dropout and the great recession, Journal of Human Capital, V. 11, 1, pp. 35–71, **TD No. 970 (October 2014).**
- ALBERTAZZI U., M. BOTTERO and G. SENE, Information externalities in the credit market and the spell of credit rationing, Journal of Financial Intermediation, v. 30, pp. 61–70, TD No. 980 (November 2014).
- ALESSANDRI P. and H. MUMTAZ, *Financial indicators and density forecasts for US output and inflation*, Review of Economic Dynamics, v. 24, pp. 66-78, **TD No. 977 (November 2014).**
- BARBIERI G., C. ROSSETTI and P. SESTITO, *Teacher motivation and student learning*, Politica economica/Journal of Economic Policy, v. 33, 1, pp.59-72, **TD No. 761 (June 2010).**
- BENTIVOGLI C. and M. LITTERIO, Foreign ownership and performance: evidence from a panel of Italian firms, International Journal of the Economics of Business, v. 24, 3, pp. 251-273, **TD No. 1085** (October 2016).
- BRONZINI R. and A. D'IGNAZIO, *Bank internationalisation and firm exports: evidence from matched firmbank data*, Review of International Economics, v. 25, 3, pp. 476-499 TD No. 1055 (March 2016).
- BRUCHE M. and A. SEGURA, *Debt maturity and the liquidity of secondary debt markets*, Journal of Financial Economics, v. 124, 3, pp. 599-613, **TD No. 1049 (January 2016).**
- BURLON L., *Public expenditure distribution, voting, and growth,* Journal of Public Economic Theory,, v. 19, 4, pp. 789–810, **TD No. 961 (April 2014).**
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Macroeconomic effectiveness of non-standard monetary policy and early exit. a model-based evaluation, International Finance, v. 20, 2, pp.155-173, TD No. 1074 (July 2016).
- BUSETTI F., *Quantile aggregation of density forecasts,* Oxford Bulletin of Economics and Statistics, v. 79, 4, pp. 495-512, **TD No. 979 (November 2014).**
- CESARONI T. and S. IEZZI, *The predictive content of business survey indicators: evidence from SIGE,* Journal of Business Cycle Research, v.13, 1, pp 75–104, **TD No. 1031 (October 2015).**
- CONTI P., D. MARELLA and A. NERI, Statistical matching and uncertainty analysis in combining household income and expenditure data, Statistical Methods & Applications, v. 26, 3, pp 485–505, TD No. 1018 (July 2015).
- D'AMURI F., *Monitoring and disincentives in containing paid sick leave*, Labour Economics, v. 49, pp. 74-83, **TD No. 787 (January 2011).**
- D'AMURI F. and J. MARCUCCI, *The predictive power of google searches in forecasting unemployment,* International Journal of Forecasting, v. 33, 4, pp. 801-816, **TD No. 891 (November 2012).**
- DE BLASIO G. and S. POY, *The impact of local minimum wages on employment: evidence from Italy in the* 1950s, Journal of Regional Science, v. 57, 1, pp. 48-74, **TD No. 953 (March 2014).**
- DEL GIOVANE P., A. NOBILI and F. M. SIGNORETTI, Assessing the sources of credit supply tightening: was the sovereign debt crisis different from Lehman?, International Journal of Central Banking, v. 13, 2, pp. 197-234, TD No. 942 (November 2013).
- DEL PRETE S., M. PAGNINI, P. ROSSI and V. VACCA, Lending organization and credit supply during the 2008–2009 crisis, Economic Notes, v. 46, 2, pp. 207–236, TD No. 1108 (April 2017).
- DELLE MONACHE D. and I. PETRELLA, Adaptive models and heavy tails with an application to inflation forecasting, International Journal of Forecasting, v. 33, 2, pp. 482-501, TD No. 1052 (March 2016).

- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, v. 40, 10, pp. 2078-2096, **TD No. 877 (September 2012).**
- GIACOMELLI S. and C. MENON, Does weak contract enforcement affect firm size? Evidence from the neighbour's court, Journal of Economic Geography, v. 17, 6, pp. 1251-1282, TD No. 898 (January 2013).
- LOBERTO M. and C. PERRICONE, *Does trend inflation make a difference?*, Economic Modelling, v. 61, pp. 351–375, **TD No. 1033 (October 2015).**
- MANCINI A.L., C. MONFARDINI and S. PASQUA, *Is a good example the best sermon? Children's imitation of parental reading*, Review of Economics of the Household, v. 15, 3, pp 965–993, **D No. 958** (April 2014).
- MEEKS R., B. NELSON and P. ALESSANDRI, *Shadow banks and macroeconomic instability*, Journal of Money, Credit and Banking, v. 49, 7, pp. 1483–1516, **TD No. 939 (November 2013).**
- MICUCCI G. and P. ROSSI, *Debt restructuring and the role of banks' organizational structure and lending technologies*, Journal of Financial Services Research, v. 51, 3, pp 339–361, **TD No. 763 (June 2010).**
- MOCETTI S., M. PAGNINI and E. SETTE, *Information technology and banking organization*, Journal of Journal of Financial Services Research, v. 51, pp. 313-338, **TD No. 752 (March 2010).**
- MOCETTI S. and E. VIVIANO, *Looking behind mortgage delinquencies*, Journal of Banking & Finance, v. 75, pp. 53-63, **TD No. 999 (January 2015).**
- NOBILI A. and F. ZOLLINO, A structural model for the housing and credit market in Italy, Journal of Housing Economics, v. 36, pp. 73-87, **TD No. 887 (October 2012).**
- PALAZZO F., Search costs and the severity of adverse selection, Research in Economics, v. 71, 1, pp. 171-197, **TD No. 1073 (July 2016).**
- PATACCHINI E. and E. RAINONE, Social ties and the demand for financial services, Journal of Financial Services Research, v. 52, 1–2, pp 35–88, TD No. 1115 (June 2017).
- PATACCHINI E., E. RAINONE and Y. ZENOU, *Heterogeneous peer effects in education*, Journal of Economic Behavior & Organization, v. 134, pp. 190–227, **TD No. 1048 (January 2016).**
- SBRANA G., A. SILVESTRINI and F. VENDITTI, *Short-term inflation forecasting: the M.E.T.A. approach,* International Journal of Forecasting, v. 33, 4, pp. 1065-1081, **TD No. 1016 (June 2015).**
- SEGURA A. and J. SUAREZ, *How excessive is banks' maturity transformation?*, Review of Financial Studies, v. 30, 10, pp. 3538–3580, **TD No. 1065 (April 2016).**
- VACCA V., An unexpected crisis? Looking at pricing effectiveness of heterogeneous banks, Economic Notes, v. 46, 2, pp. 171–206, TD No. 814 (July 2011).
- VERGARA CAFFARELI F., One-way flow networks with decreasing returns to linking, Dynamic Games and Applications, v. 7, 2, pp. 323-345, **TD No. 734 (November 2009).**
- ZAGHINI A., A Tale of fragmentation: corporate funding in the euro-area bond market, International Review of Financial Analysis, v. 49, pp. 59-68, **TD No. 1104 (February 2017).**

#### 2018

- ADAMOPOULOU A. and E. KAYA, *Young Adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, **TD No. 1038 (November 2015).**
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, TD No. 1147 (October 2017).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, TD No. 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, **TD No. 1066 (March 2018).**
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, TD No. 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, **TD No. 1030 (October 2015).**

- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, **TD No. 1077 (July 2016).**
- CECCHETTI S., F. NATOLI and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring*, International Journal of Central Banking, v. 14, 1, pp. 35-71, **TD No. 1025 (July 2015).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, **TD No. 1063 (April 2016).**
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, **TD No. 1054 (March 2016).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, TD No. 1100 (February 2017).

#### FORTHCOMING

- ALBANESE G., G. DE BLASIO and P. SESTITO, *Trust, risk and time preferences: evidence from survey data,* International Review of Economics, **TD No. 911 (April 2013).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, **TD No. 1098 (February 2017).**
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces,* Regional Science and Urban Economics, **TD No. 1057 (March 2016).**
- BELOTTI F. and G. ILARDI, Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, **TD No. 1147 (October 2017).**
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, **TD No. 1097 (February 2017).**
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, **TD No. 909 (April 2013).**
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, **TD No. 1117 (June 2017).**
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, **TD No. 985 (November 2014).**
- D'AMURI F., Monitoring and disincentives in containing paid sick leave, Labour Economics, TD No. 787 (January 2011).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, **TD No. 1174 (April 2018).**
- FEDERICO S. and E. TOSTI, *Exporters and importers of services: firm-level evidence on Italy*, The World Economy, **TD No. 877 (September 2012).**
- GERALI A. and S. NERI, *Natural rates across the atlantic,* Journal of Macroeconomics, **TD No. 1140** (September 2017).
- GIACOMELLI S. and C. MENON, *Does weak contract enforcement affect firm size? Evidence from the neighbour's court,* Journal of Economic Geography, **TD No. 898 (January 2013).**
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, TD No. 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, **TD No. 1025 (July 2015).**
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, **TD No. 871 (July 2012).**
- SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, TD No. 1100 (February 2017).